

Single-item dynamic lot-sizing problems: An updated survey

Nadjib Brahimi, Nabil Absi, Stéphane Dauzère-Pérès, Atle Nordli

▶ To cite this version:

Nadjib Brahimi, Nabil Absi, Stéphane Dauzère-Pérès, Atle Nordli. Single-item dynamic lot-sizing problems: An updated survey. European Journal of Operational Research, 2017, 263 (3), pp.838-863. 10.1016/j.ejor.2017.05.008 . emse-01792227

HAL Id: emse-01792227 https://hal-emse.ccsd.cnrs.fr/emse-01792227v1

Submitted on 13 Feb 2020 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Single-item dynamic lot-sizing problems: An updated survey

Nadjib Brahimi, Nabil Absi, Stéphane Dauzère-Pérès, Atle Nordli

▶ To cite this version:

Nadjib Brahimi, Nabil Absi, Stéphane Dauzère-Pérès, Atle Nordli. Single-item dynamic lot-sizing problems: An updated survey. European Journal of Operational Research, Elsevier, 2017, 263 (3), pp.838-863. 10.1016/j.ejor.2017.05.008. emse-01792227

HAL Id: emse-01792227 https://hal-emse.ccsd.cnrs.fr/emse-01792227

Submitted on 13 Feb 2020 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Single-Item Dynamic Lot-Sizing Problems: An Updated Survey

Nadjib Brahimi¹ Nabil Absi² Stéphane Dauzère-Pérès^{2,3} Atle Nordli³

¹ Department of Supply Chain Management Rennes School of Business F-35065 Rennes, France E-mail: nadjib.brahimi@esc-rennes.com

²Department of Manufacturing Sciences and Logistics, CMP Ecole des Mines de Saint-Etienne, CNRS UMR 6158 LIMOS, F-13541 Gardanne, France E-mail: absi@emse.fr, dauzere-peres@emse.fr

³Department of Accounting, Auditing and Business Analytics BI Norwegian Business School, 0484 Oslo, Norway Email: atle.nordli@bi.no

Abstract

Following our previous paper (Brahimi, N., Dauzère-Pérès, S., Najid, N. M., and Nordli, A. Single item lot sizing problems. European Journal of Operational Research, 168(1):1-16, 2006), we present an updated and extended survey of Single-Item Lot-Sizing Problems with focus on publications from 2004 to 2016. Exact and heuristic solution procedures are surveyed. A concise and comprehensive summary of different extensions of the problem is given. The classification of the extensions is based on different characteristics such as resource limitations, assumptions on demand and cost structure. The large number of surveyed papers shows the increased interest of researchers in lot-sizing problems in general and in single-item problems in particular. The survey and the proposed classification should help researchers to identify new research topics, to propose relevant problems and/or novel solution approaches.

Keywords: Production; Planning; Lot Sizing; Single Item; Dynamic Demand

1. Introduction

Since the publication of the seminal work of Wagner and Whitin (1958) in 1958, a lot of research has been conducted on the Single-Item Lot-Sizing Problem (SILSP) and its extensions. The SILSP is interesting in itself to model some tactical production and distribution planning problems, e.g. when planning the replenishment of one raw material with fixed and variable ordering and transportation costs. It is also important to efficiently solve the SILSP because it appears as a subproblem in the solution procedures of many complex lot-sizing problems such as capacitated multi-item problems. Extensions of the SILSP include production capacity, remanufacturing, backlogging, lost sales, demand and production time windows, bounded inventory, perishability, etc. The large number of publications on the SILSP and the large variety of its applications and extensions call for a literature review to put together all the relevant references and classify them. This will allow researchers in the field to more easily identify research trends and gaps to be filled in lot sizing in particular and in production and distribution planning in general. This survey is an extension and an update of a previous survey published by three of the authors in *European Journal of Operational Research* in 2006 (Brahimi et al. (2006)). Since this survey was published, there has been a growing interest in the study of the SILSP. There were at least 100 publications on the SILSP and its extensions in the past 8 years. The objective of this paper is to update the previous literature review and enrich it with the new research in the area. To avoid repetition and for space reasons, most of the references and details in (Brahimi et al. (2006)) are omitted.

The SILSP can be defined as a planning problem in which there is time-varying demand for a single product over a planning horizon of T periods. The objective is to determine periods where production will take place and the quantities that have to be produced in these periods. The total production should satisfy the demands while minimizing the total costs. The basic costs are the unit production cost p_t (where t = 1, ..., T is the period); the setup cost s_t , which is a fixed cost incurred if a production process is started in a period t, and the unit inventory holding cost h_t . Extensions might include, for example, a capacity limitation C_t or inventory bound I^{max} .

The classification of lot-sizing problems can be done based on several criteria or characteristics such as: Nature of data (deterministic or stochastic), nature of the time scale (continuous or discrete), number of machines, number of production stages (levels), capacity constraints and their nature (fixed or variable), length of production periods, etc. In addition to the classifications proposed in (Brahimi et al. (2006), in their book, Pochet and Wolsey (2006) discuss models for different lot sizing and production planning problems.

This paper focuses on discrete time models. For surveys on continuous time models, we invite the reader to consult Holmbom and Segerstedt (2014), for example. There are also some surveys on discrete time lot-sizing problems. The majority of these surveys deal with multi-item and multi-level problems, some of which are cited here in chronological order: Bahl et al. (1987), Karimi et al. (2003), Pochet and Wolsey (2006), Jans and Degraeve (2008), Buschkühl et al. (2010), Diaz-Madroñero et al. (2014).

In this paper, we restrict ourselves to the single-item dynamic lot-sizing problem and its extensions. This survey is an update of previously published survey (Brahimi et al. (2006)). It is motivated by several recent extensions of the single-item lot-sizing problem such as: Remanufacturing, stochastic versions, bounded inventories, carbon emission constraints, minimum order quantities, batch sizes, etc. We do not claim to have an exhaustive list of papers in the area of the single-item lot-sizing problem and its extensions. Since the number of references in this area is huge, we apologize in advance if we missed any relevant publication. Through this work we wanted to give a general overview of studied problems and their importance. Papers are not classified in a chronological order but cited when needed in the closely linked section.

By looking closely at all surveyed papers since 1958, we found that more than 50% of the papers were published during the last 10 years. Furthermore, the main journals publishing on this topic, totaling more than 50% of all publications, are *European Journal of Operational Research*, *Management Science*, *Operations Research*, *International Journal of Production Economics*, and *International Journal of Production Research*.

The paper is organized as follows. Section 2 deals with basic formulations of the classical singleitem lot-sizing problem. Section 3 gives a quick overview of used solution methods. In Section 4 we propose a classification of single-item lot-sizing problems and a quantitative analysis of the bibliography. Section 5 deals with papers addressing assumption on demands such as backlogging, lost sales, time windows, stochastic demand, elastic demand and profits. In Section 6, we address resource constraints such as capacity, inventory, lot-sizes constraints. Section 7 regroups single-item lot-sizing problems with complex structures like multi-level, multi-echelon and remanufacturing structures. Section 8 addresses single-item lot-sizing problems that integrate other decision levels such as scheduling, warehouse location, transportation, vehicle routing, etc. Section 9 gathers several extensions such as pricing, cost structures, co-production, environmental issues, stochastic parameters, etc. The last section concludes the paper and provides some promising research directions.

2. Formulations of the basic problem

Traditionally, Mixed Integer Programming (MIP) formulations are not directly used to solve the uncapacitated SILSP. But, as already mentioned, the SILSP is very often solved as a sub-problem in several algorithms for more complex lot-sizing problems, which are often modeled as Mixed Integer Programs. This is why we present the different MIP formulations of the uncapacitated SILSP. Also, some authors derive Dynamic Programming (DP) algorithms using some structural properties of these formulations.

First, we introduce a general model that does not make any assumptions on the shape of production and inventory holding costs, then we present straightforward compact formulation that provides a weak linear relaxation. We then briefly discuss two reformulations and an extended formulation with the so-called (l, S)-inequalities.

Let T be the length of the planning horizon and d_t be the deterministic demand in period t (t = 1, ..., T). $f_t^p(.)$ and $f_t^h(.)$ are respectively the production cost and the holding cost functions at period t (t = 1, ..., T). The goal is to decide when to produce and how much to produce in order to satisfy demands and minimize the total production and holding costs. The decision variables are: X_t , the quantity to be produced in period t and I_t , the inventory level at the end of period t (t = 1, ..., T). We assume, without loss of generality, that the stock at the beginning and the stock at the end of the planning horizon are zero.

2.1. A general model

The most general formulation of the uncapacitated SILSP that does not make any assumptions on the structure of functions $f_t^p(.)$ and $f_t^h(.)$ can be written as follows:

Minimize
$$\sum_{t=1}^{T} (f_t^p(X_t) + f_t^h(I_t))$$
(1)

Subject to:

$$I_{t-1} + X_t = d_t + I_t \qquad \forall t \qquad (2)$$
$$I_t, X_t \ge 0 \qquad \forall t \qquad (3)$$

The objective function (1) minimizes the total production and holding costs over the horizon of T periods. Constraints (2) are the inventory balance equations. They express that the entering stock (I_{t-1}) added to the current period production (X_t) are used to satisfy the demand (d_t) , and what remains is kept in stock at the end of the period (I_t) . Constraints (3) define the continuous production variables X_t and inventory levels I_t .

In this section, we limit the discussion to functions $f_t^p(.)$ and $f_t^h(.)$ of the form: $f_t^p(x_t) = s_t \delta(x(t)) +$ $p_t x_t$ and $f_t^h(I_t) = h_t I_t$, where p_t is the unit production cost in each period t, h_t is the unit holding cost in each period t, s_t is the fixed setup cost incurred once in a period if production occurs, and $\delta(x(t)) = 1$ if $x_t > 0$ and zero otherwise. This is actually the most frequently encountered form of these functions in the lot-sizing literature. Moreover, there is a special case where the unit production and holding costs satisfy the condition $p_{t-1} + h_{t-1} \ge p_t$ for all periods, i.e. it is always optimal to produce as late as possible (ignoring setup costs). In the literature, this is often referred as the case without speculative motives to hold inventory or the case with Wagner-Whitin costs. The uncapacitated SILSP with no speculative motives is known to be solvable in O(T). Note that there are many practical instances having such cost structure (Wolsey (1995)). Finally, and to simplify the presentation, we replace function $\delta(x(t))$ with a binary decision variable Y_t .

2.2. Aggregate formulation (AGG)

Let $d_{\tau t} = d_{\tau} + d_{\tau+1} + \dots + d_t$. Using the simplified production and holding cost functions, the following aggregate (AGG) formulation can be written:

Minimize
$$\sum_{t=1}^{T} (s_t Y_t + p_t X_t + h_t I_t)$$
(4)

Subject to:

$$I_{t-1} + X_t = d_t + I_t \qquad \forall t \tag{5}$$

$$\begin{aligned} X_t &\leq Y_t d_{tT} & \forall t & (6) \\ Y_t &\in \{0, 1\} & \forall t & (7) \end{aligned}$$

$$0,1\} \qquad \forall t \qquad (7)$$

$$I_t, X_t \ge 0 \qquad \qquad \forall t \tag{8}$$

This formulation is called *aggregate* in contrast with disaggregated formulations such as facility location problem based formulation and the shortest path problem based formulation, to which we refer as FAL and SHP, respectively. In FAL and SHP formulations production variables are split and presented in a less intuitive way than the variables X_t above. The objective function (4) minimizes the sum of the setup, production and inventory holding costs. Constraints (5) are the inventory balance equations. Constraints (6) relate the continuous production variables X_t to the binary setup variables Y_t . Constraints (7) and (8) define the decision variables.

In the above formulation, using the fact that $I_t = \sum_{\tau=1}^t X_{\tau} - \sum_{\tau=1}^t d_{\tau}$, t = 1, ...T and combining this with constraint $I_t \ge 0 \quad (\forall t)$ and constraints (5), one can derive a formulation without inventory variables.

2.3. Strong Formulations

FAL and SHP formulations (see Brahimi et al. (2006) and the references therein for details) are considered as tight formulations because their LP relaxations have optimal solutions in which the variables Y are integer. Some fast dynamic programming algorithms were drived based on these formulations in the 1990s.

FAL and SHP are reformulations of the problem as the original variables (production variables in this case) are replaced with new variables. By adding valid inequalities to AGG formulation, it is possible to develop strong formulations with the original decision variables. Barany et al. (1984) introduced (l, S) inequalities defined as:

$$\sum_{t\in S} X_t + \sum_{t\in \bar{S}} d_{tl} Y_t \ge d_{1l} \tag{9}$$

where $l \in \{1, \ldots, T\}$, $S \subseteq \{1, \ldots, l\}$ and $\overline{S} = \{1, \ldots, l\} \setminus S$.

Barany et al. (1984) showed that combining (4)-(8) with the (l, S)-inequalities (9) provides a complete polyhedral description of the convex hull of the uncapacitated SILSP. The new formulation would contain an exponential number of valid inequalities. However, they can be used in a cutting plane approach to solve the problem.

3. Complexity and solution approaches

Most of the literature of the basic SILSP and its extensions has been on exact solution methods. Such methods include dynamic programming, polyhedral approaches, branch-and-cut and branch-andbound algorithms. Nevertheless, heuristic methods such as simple construction heuristics, approximation algorithms, and some improvement heuristics have also been developed. These have played an important role in developing heuristics for more complex lot-sizing problems.

It is worth mentioning that it is possible to derive a closed-form expression for the optimal solution of the very special case of the SILSP where demand is constant (Ganas and Papachristos (2005)). Some studies focused on classifying SILSPs based on their computational complexity.

In what follows, we start with the presentation of the complexity of the SILSP; then we present the most popular solution methods, first exact methods and then heuristics.

3.1. Complexity

The basic uncapacitated SILSP is an "easy" problem that can be solved in $O(T \log T)$ in general and in O(T) with some assumptions, in particular without speculative motives to hold inventory. However, some of the extensions of the SILSP are NP-hard. In addition to complexity studies presented in Brahimi et al. (2006) on the capacitated SILSP, recent studies include Akbalik and Rapine (2013) for the problems with constraints on batch sizes, and Guan (2011) for stochastic problems. Details about these studies will be presented in the corresponding section of each extension.

3.2. Dynamic Programming

The first exact algorithm developed for the uncapacitated SILSP is the dynamic programming algorithm proposed by Wagner and Whitin (1958) (WW). It is a forward dynamic programming algorithm that runs in $O(T^2)$. Using the the Zero Inventory Ordering (ZIO) property, the search space is reduced to at most T(T + 1)/2.

A natural consequence of the ZIO property is: There exists an optimal solution in which if $X_t > 0$, then $X_t = \sum_{i=t}^{t+k} d_i$ for some $k \ge 0$, i.e. each production period satisfies the demands of k + 1 consecutive periods starting from period t.

Let G(k) be the minimum cost of the subproblem from periods 1 up to k. In such a solution, if t is the last period in which production occurs, then $X_t = d_{t,k}$. Thus, the recursive equation is:

$$G(k) = \min_{1 \le t \le k} \{ G(t-1) + s_t + p_t d_{t,k} + \sum_{\tau=t}^{k-1} h_\tau d_{\tau+1,k} \}$$
(10)

Then, initializing G(0) to 0 and calculating G(k) for k = 1, ..., T leads to the value G(T) of the optimal solution. This is done in $O(T^2)$ if the values G(1), G(2), ..., G(T) are evaluated in the simplest way. Finally, by starting from G(T) and going backward, the corresponding optimal solution is derived in O(T) additional time.

Some researchers attempted to improve the WW algorithm through various efficient implementations (Zabel (1964), Bahl and Taj (1991)). However, these implementations do not improve the complexity of the WW algorithm.

By the end of the 1980's, three groups of researchers independently developed three different dynamic programming algorithms each of which can solve the classical WW problem in $O(T \log T)$ (See Brahimi et al. (2006)).

We will also see in Sections 5–8 that dynamic programming is extensively used to solve different extensions of the SILSP.

3.3. Branch-and-bound methods and dual algorithms

Unlike dynamic programming, very few researchers developed Branch-and-Bound (B&B) procedures for the SILSP. For the SILSP with capacity constraint, B&B algorithms were developed by Erenguc and Tufekci (1987), Erenguc and Aksoy (1990), and Lotfi and Yoon (1994). In Chung et al. (1994), an exact algorithm that combines B&B and dynamic programming was presented. Finally, developing good bounds for capacitated SILSPs improves the performance of branch-and-bound algorithms. Hardin et al. (2007) propose procedures to quickly generate upper and lower bounds for a special class of capacitated SILSPs.

Two different efficient dual-based optimization algorithms are proposed in van Hoesel et al. (1991) and Levi et al. (2006). The algorithm in van Hoesel et al. (1991) is based on the formulation proposed by Barany et al. (1984). The primal-dual algorithm in Levi et al. (2006) is quite simple and is derived from a primal-dual algorithm proposed for the more complex joint replenishment problem.

3.4. Valid inequalities and branch-and-cut methods

The polyhedral structure of SILSPs was considerably studied in the literature. The first purpose is to directly use MIP solvers on extended formulations; that is formulations with added valid inequalities. The second purpose of these studies is to solve the problem using branch-and-cut (B&C) algorithms. A lot of research was carried out on the development of different types of valid inequalities for different classes of SILSPs. The earliest research on this topic is the paper of Barany et al. (1984) on the uncapacited SILSP. Other studies include, for example, Wolsey (1989) who presents families of strong valid inequalities for the uncapacited SILSP with startup costs. Di Summa and Wolsey (2010), Escalante et al. (2011), Küçükyavuz and Pochet (2009), Pochet and Wolsey (1993) and van Vyve (2006) develop extended formulations whose LP relaxation solves the lot-sizing problem. Pereira and Wolsey (2001) define facets and prove that the face of optimal solutions is found in $O(T^2)$. Loparic et al. (2003) use knapsack inequalities as strong valid inequalities for some capacitated SILSPs. Readers interested in this topic can refer to the book of Pochet and Wolsey (2006). Valid inequalities that are developed for SILSPs can be used to strengthen the LP relaxation of richer and harder lot-sizing problems such as those dealing with multiple items.

3.5. Simple Heuristics

The simplest way to solve the uncapacitated SILSP is the lot-for-lot procedure where production at each period is equal to the demand at that period. Another simple heuristic to the uncapacitated SILSP is based on the Economic Order Quantity (EOQ) formula. The EOQ is calculated based on the demand, holding cost and setup cost that are averaged over the T periods of the planning horizon. If the production must be positive at a given period t $(X_t > 0)$, then $X_t = EOQ$. Another alternative is to set $X_t = d_{t\tau}$ $(\tau \ge t)$ such that $d_{t,\tau}$ is the closest value to EOQ.

The heuristic proposed by Silver and Meal (1973) is a forward method that requires determining the average cost per period as a function of the number of periods whose demand is to be produced in the current period, and stopping the computation when this function first increases. The Least Unit Cost heuristic is a modified version of the Silver and Meal heuristic which minimizes the cost per unit of demand. Finally, one of the most popular heuristics is the Part Period Balancing algorithm (DeMatteis (1968), Wemmerlöv (1983)). It consists in setting the horizon (number of periods) on which to order to the number of periods that most closely matches the total holding cost with the setup cost over that period. A new heuristic was proposed in van den Heuvel and Wagelmans (2009) with a similar worst case ratio. van den Heuvel and Wagelmans (2010) proved that a worst case ratio of 2 is actually the best possible result for any myopic heuristic. Reviews and comparisons of these heuristics and others are presented in Benli et al. (1988), Nydick and Weiss (1989), Coleman (1992), Vachani (1992), and Baciarello et al. (2013).

Knowing that there exist fast exact algorithms in $O(T \log T)$ to solve the uncapacitated SILSP, one might claim that simple heuristics are of limited interest in practice, even though their complexity is usually O(T). Actually heuristics have at least three advantages. Firstly, they can be used for academic purposes to introduce students to production planning concepts. Secondly, heuristics are used in practical applications to directly solve single-item problems or complex industrial problems as they can be more easily adapted. Many heuristics are integrated into some ERP (Enterprise Resource Planning) systems (see for example Bahl and Neelam (2009)). As a third advantage, Blackburn and Millen (1980) showed that at least some of the simple heuristics (including the Silver and Meal heuristic) are more efficient than the Wagner-Whitin algorithm for rolling horizon problems. This is an important point as most real applications are dynamic in nature and new solutions have to be computed frequently as more accurate information (e.g. on demand) becomes available. However, it seems that this last advantage of heuristics is no longer relevant as Stadtler (2000) showed that exact solution procedures used with adjusted data perform at least as well as simple heuristics.

Simple heuristics are mainly used to solve harder problems such as multi-item lot-sizing problems. To the best of our knowledge, there are very few applications of these heuristics to solve the SILSP extensions presented in this paper. The relevant references will be presented in the adequate sections below.

3.6. Approximation methods

Few theoretical results have been published on approximation methods applied to uncapacitated single-item lot-sizing problems. As discussed in Section 3.5, van den Heuvel and Wagelmans (2009) and van den Heuvel and Wagelmans (2010) provide worst case error bounds for different heuristics for the uncapacitated SILSP.

Fully polynomial approximation schemes for the capacitated SILSP are discussed in Section 6.1.2 and in Section 9.1 for the stochastic SILSP.

3.7. Other heuristics

The single-item lot-sizing problem is considered as a relatively "easy" problem since most of its difficult variants are NP-hard in the ordinary sense, and can be solved with pseudo-polynomial time algorithms. This can explain why there are very few publications on the application of advanced heuristics such as meta-heuristics to solve these problems. Hence, the few advanced heuristics that can be met are mostly developed for extensions of the SILSP. For example, Lagrangian relaxation-based heuristics were developed by Zhang et al. (2012) and Brahimi and Dauzère-Pérès (2015) to solve the capacitated SILSP with remanufacturing and production time windows, respectively. Tabu search was used by Armentano et al. (2011) on an integrated lot-sizing and distribution problem. Parsopoulos et al. (2015) investigate the performance of Differential Evolution algorithm on the uncapacitated SILSP with remanufacturing. Finally, goal programming was used by Choudhary and Shankar (2014) to solve a SILSP with multiple suppliers.

4. Classification

Table 1: A classification of lot-sizing problems						
Parameter	Classifications					
Information degree	Deterministic [*] , stochastic [*] .					
Horizon	finite*, infinite.					
Time scale	discrete (small time periods, large time periods [*]), continuous					
Number of items	single item [*] , multi item.					
Number of levels	single level [*] , multi level [*] (serial, in-tree, general,).					
Relevant costs	setup related (startup*, reservation*), inventory related (holding*, back-					
	order [*] , lost sales [*]),					
	capacity related (regular hours [*] , overtime [*] , sub-contracting [*]).					
Resource constraints	number (single resource [*] , multi resource [*]), type (constant [*] , variable [*]).					
Service Policy	demand satisfied on-time*, backorder*, lost sales*, sub-contracting*.					
Time consuming activi-	setup time (ST) (minor ST , major ST), processing time (zero, con-					
ties	stant [*] , variable), lead time, transportation time.					
Objectives	minimize costs [*] , maximize service level, smoothing of production load,					
	maximize profit*					

The classification of lot-sizing problems can be based on several criteria such as those presented in Haase (1994) and Brahimi (2004). These criteria are summarized in Table 1.

(*) Classification corresponding to the problems considered in this survey.

Inspired by this classification, the papers in this survey were grouped by category. Since a large part of the studies deals with assumptions on demands (such as backlogging, lost sales, time windows, stochastic demand, elastic demand and profits), the related papers are reviewed in Section 5. Other important extensions deal with resource constraints (such as capacity, inventory, lot-size constraints). Articles dealing with this topic are grouped in Section 6. Generally SILSPs with complex structures (such as multi-level, multi-echelon and remanufacturing) are NP-Hard. Papers dealing with these extensions are analyzed in Section 7. Lot-Sizing problems combined with other decisions (such as scheduling, warehouse location, transportation, vehicle routing, etc.) are considered in Section 8. In Section 9, we grouped all remaining relevant extensions (such as pricing, cost structures, co-production, environmental issues, stochastic parameters, etc.) that do not fit within one of the previous categories.

5. Assumptions on demands

As mentioned in the previous sections, several assumptions are made for the classical single-item uncapacitated lot-sizing problem. The strongest assumptions are related to demand. In fact, usually the demand is supposed deterministic and should be satisfied entirely on time without having the possibility to lose it or to postpone it to later periods. In real-life problems, these assumptions are not realistic in many cases. For example, generally when the capacity is not sufficient to satisfy the entire demand, several options can be considered. The demand can be backlogged to future periods or partially/totally lost. The demand can also be considered as a stochastic variable following a given distribution or represented by a scenario tree. Some studies addressed profit maximization rather than demand satisfaction, while others considered elastic demands (i.e. demand is a variable). In this section, we survey the papers addressing these assumptions.

5.1. Backlogging

Backlogging is one of the first extensions of the classical uncapacitated SILSP. The SILSP with backlogging was first studied by Zangwill (1969). Backlogging can be due to two situations, either the cost of making and storing the product is not profitable or capacity constraints are not sufficient to satisfy demand on time. In both situations, the backlogged quantities are modeled using new flow variables that are in the opposite direction of inventory flow variables. More formally, to generalize the classical uncapacitated SILSP, we introduce a new non negative continuous variable Z_t that represents the accumulated backlog at the end of period t. A backlogging cost b_t is associated with each unit backlogged from period t-1 to period t. The classical model for the uncapacitated SILSP can be easily generalized by replacing Constraints (5) with Constraints (11), adding the cost component $\sum_{t=1}^{T} b_t Z_t$ to the objective function and introducing the non negative variable Z_t with $Z_0 = 0$. The SILSP with backlogging is denoted SILSP-B.

$$I_{t-1} - Z_{t-1} + X_t = d_t + I_t - Z_t \ \forall t \in \{1, \dots, T\}$$
(11)

Similarly to the uncapacitated SILSP, the extreme optimal solutions have the structure of a spanning tree. Zangwill (1969) shows that the problem can be solved optimally in $O(T^2)$ by generalizing the algorithm of Wagner and Whitin (1958). Analogously to the uncapacitated SILSP, this dynamic programming recursion can be represented as a shortest path problem (see Pochet and Wolsey (2006)). Using techniques similar to the ones used for the uncapacitated SILSP, the complexity was reduced to $O(T \log T)$ by several authors (see Brahimi et al. (2006)). This complexity reduces to O(T) when there are no speculative motives for late production (see Section 2.1 for inventory), i.e. if both $p_{t-1} + h_{t-1} \ge p_t$ and $p_t + b_{t-1} \ge p_{t-1}$ for all periods.

In a recent study, Küçükyavuz and Pochet (2009) identify valid inequalities that subsume all previously known valid inequalities for the uncapacitated SILSP-B. They show that these inequalities are enough to describe the convex hull of solutions.

Several studies extended the uncapacitated SILSP-B and showed that the problem generally remains polynomially solvable. Absi et al. (2011) extend the uncapacitated SILSP with production time windows by considering backlogging costs, and show that the problem can be solved optimally with an $O(T^2)$ dynamic programming algorithm. For the uncapacitated SILSP-B with demand time windows, Hwang (2007) proposes an $O(T^3)$ dynamic programming algorithm to solve the problem under the nonspeculative cost structure. For the general cost structure, he proposes an $O(\max[T^2, nT])$ algorithm,

ACCEPTED MANUSCRIPT

where n is the number of time windows with positive demands. Chu et al. (2013) generalize the uncapacitated SILSP-B by considering outsourcing and inventory capacity. The backlogging level at each period is supposed to be limited. The authors show that this problem can be solved in $O(T^4 \log T)$. van Vyve (2007) proposes an $O(T^3)$ algorithm to optimally solve the single-item lot-sizing problem with constant batch size and backlogging. van Vyve (2006) presents two linear-programming extended formulations of the constant-capacity lot-sizing problem with backlogging. The first one addresses the problem with a general cost function and has $O(T^3)$ variables and constraints. The second deals with the problem when there are no speculative motives and has $O(T^2)$ variables and $O(T^3)$ constraints. An $O(T^3)$ algorithm is proposed by Ou (2012) to solve a SILSP-B with time independent capacity.

Generally, backlogging costs are not easy to evaluate. In fact, in addition to the penalties due to late deliveries several other costs should be considered. The loss of customer goodwill is one of these components and it is not easy to quantify (Aksen (2007)). Additional costs related to the expedition of the late demand could be considered.

5.2. Lost sales

Lost sales is generally considered as an alternative to backlogging but they can also be considered together. In fact, a demand can be postponed for a given number of periods, but if the delivery date is too late the demand can be lost. Similarly to backlogging, lost sales can be due to two situations. Either capacity constraints are not sufficient to fulfill the demand or the cost of producing and storing the product is not profitable. The model should decide which demand to satisfy and which demand to lose. Aksen et al. (2003) proposed a $O(T^2)$ algorithm to solve this problem. To generalize the classical SILSP, a new non negative continuous variable R_t is introduced that represents the unmet demand at the end of each period t. A unitary lost sales cost l_t is associated with each unit of lost sales at period t. The classical model for the uncapacitated SILSP can easily be generalized by replacing inventory balance constraints (Constraints (5)) with constraints $I_{t-1} + X_t + R_t = d_t + I_t \ \forall t \in \{1, \ldots, T\}$, adding the cost component $\sum_{t=1}^{T} l_t R_t$ to the objective function and introducing the non negative lost sales variable R_t . The SILSP with lost sales is denoted by SILSP-LS.

The uncapacitated SILSP-LS can be represented using a fixed charge network. Similarly to the uncapacitated SILSP, the extreme optimal solutions have the structure of a spanning tree (see Aksen et al. (2003) for more details). Loparic et al. (2001) studied the uncapacitated SILSP with sales instead of lost sales and lower bounds on stocks. The objective is to maximize the income of sales rather than minimizing the cost of lost sales. They showed that the problem can be solved in polynomial time using dynamic programming. They also provide two extended formulations as well as a complete description of the convex hull of solutions.

Several studies extended the uncapacitated SILSP-LS and showed that the problem generally remains polynomially solvable. Absi et al. (2011) extend the uncapacitated SILSP with production time windows by considering lost sales costs. They show that the problem can be solved optimally with an $O(T^2)$ dynamic programming algorithm. Berk et al. (2008) study the uncapacitated SILSP with capacity constraints and lost sales for a warm/cold process. They also used dynamic programming to solve this problem. Recently, several authors studied the uncapacitated SILSP-LS with bounded inventory (Liu et al. (2007), Hwang et al. (2013), Liu and Tu (2008)). Some authors addressed outsourcing which can be modeled as lost sales (Chu and Chu (2007), Chu et al. (2013)). SILSPs with outsourcing are surveyed in Section 6.4. Evaluating the cost of lost sales is even harder than estimating the backlog cost. In fact, in addition to the loss of revenue, this cost should integrate the loss of customer goodwill. The cost of lost sales is not easy to evaluate since it should represent the impact of a low service level on future demands. If lost sales lead to outsourcing, the related cost is easier to estimate since it represents the loss of profit. How to estimate backlog and lost sales costs is not well addressed in the literature except for few publications (e.g. Oral et al. (1972), Liberopoulos et al. (2010)).

5.3. Time windows

Lot-sizing problems with time windows were introduced by Lee et al. (2001) for demand or delivery time windows and by Dauzère-Pérès et al. (2002) for production time windows.

The *demand/delivery time windows* problem in Lee et al. (2001) is characterized by the fact that demand time windows are fixed by customers and considered as grace periods during which demand can be satisfied with no penalty; i.e. no inventory or backlogging costs are incurred when demands are completed within their time windows. Such a situation can be seen in third party logistics and vendor managed inventory settings (e.g. Jaruphongsa et al. (2004)).

Lee et al. (2001) assume special conditions on costs and study two cases: With and without backlogging. For the no-backlogging problem, an $O(T^2)$ algorithm is proposed. When backlogging is allowed, the problem is solved in $O(T^3)$. For problems with more general structures, Hwang (2007) proposes a fast algorithm in $O(\max[T^2, nT])$ where n is the number of time windows.

Dauzère-Pérès et al. (2002) show the importance of production time windows by considering constraints on the availability of demands with different applications such as remanufacturing (van den Heuvel and Wagelmans (2008)), bounded inventory (Wolsey (2006)), major and minor demands (Hwang and Jaruphongsa (2008)), and raw material availability (Brahimi et al. (2015)). The SILSP with production time windows consists of processing customer demands which are not necessarily available at the first period of the planning horizon. A time window demand $d_{t1,t2}$ is characterized by the fact that production cannot start before its release period t1 and must be delivered not later than its due date t2. Besides the general case called customer specific (CS) problem, a special case called the non-customer specific (NCS) problem is distinguished.

Dauzère-Pérès et al. (2002) also identified some interesting properties of the problem and suggested dynamic programming algorithms to solve the uncapacitated case using an exponential time algorithm for the CS problem and an $O(T^4)$ algorithm for the NCS problem. Later, Wolsey (2006) further analyzed the two cases and proposed improved algorithms, in particular, an $O(T^2)$ algorithm for the NCS problem. The CS problem was also solved by Hwang (2007) using an $O(T^4)$ algorithm. The equivalence between lot-sizing problems with production time windows, the lot-sizing problem with bounded inventory, the lot-sizing problem with remanufacturing options, and the lot-sizing problem with cumulative capacities is discussed in van den Heuvel and Wagelmans (2008). Extensions of the SILSP with production time windows include backlogging, lost sales and early production (Absi et al. (2011)) and production capacity (Brahimi and Dauzère-Pérès (2015)).

5.4. Others: Stochastic and elastic demands

Most of the lot-sizing literature focuses on problems with deterministic demands. However, production planning decisions are often based on forecasts which may contain errors that affect the solution procedure to use. Thus, considering demand as a stochastic parameter can be more realistic in practice. In addition to demand, other parameters were also considered as stochastic in different studies. Lot-sizing models with different stochastic parameters (especially demand and costs) are surveyed in Section 9.1.

In lot-sizing models with elastic demand, the demand is a function of the unit price of the product. Thus, the unit price is a decision variable to be determined, demand is not known in advance, and can be increased or decreased by varying the unit price. More details on this problem are given in Section 9.2.

6. Constraints on resources

The assumption that there is an infinite amount of available resources is not realistic in many practical cases. Thus new constraints need to be added to uncapacitated SILSPs to handle scarcity of machine/worker time, storage space, and technological characteristics of some resources (such as the maximum amount that can be handled by a machine at a time).

6.1. Production capacity constraints

In most practical situations, capacity cannot be assumed infinite. Capacitated SILSPs are harder than their uncapacitated counterparts. However, there are some "easy" cases. Most NP-completeness results and first polynomial time algorithms for capacitated SILSPs were presented in Bitran and Yanasse (1982) who proposed a four-field notation to classify these problems. With each problem is associated a quadruple $\alpha/\beta/\gamma/\delta$, where α , β , γ and δ specify the special structures of the setup cost, unitary holding cost, unitary production cost and capacity, respectively. The values taken by α , β , γ and δ are G, C, ND, NI and Z which correspond to General structure, Constant, Non-Decreasing, Non-Increasing and Zero, respectively. For example, the notation NI/ND/C/G indicates a family of problems where, over time, the setup cost s_t is non-increasing, the holding cost h_t is non-decreasing, the production cost p_t is constant and the capacity C_t is not restricted to any pre-specified structure.

6.1.1. Polynomial cases

The complexity of the capacitated SILSP mainly depends on the capacity parameter structure (variable or time independent). The following problems were solved in polynomial times either in Bitran and Yanasse (1982) or in other publications until 2005 (See Brahimi et al. (2006))): NI/G/NI/ND, NI/G/NI/C, C/Z/C/G, ND/Z/ND/NI, G/G/G/C. Algorithms in $O(T \log T)$ were proposed to solve Z/G/G/G and NI/G/G/C problems by Kovalyov and Pesch (2014) and Feng et al. (2011), respectively.

New polynomial algorithms were developed for extensions or special cases of capacitated SILSPs, for instance with stepwise production cost or batch production (see Section 6.3.2), with backlogging (see Section 5.1), or with reservation costs (see Section 9.4).

6.1.2. Other results and solution approaches

In general, the capacitated SILSP is NP-hard even for the following special cases (Florian et al. (1980) and Bitran and Yanasse (1982)): C/Z/NI/NI, C/Z/ND/ND, ND/Z/Z/ND, NI/Z/Z/NI, C/G/Z/NI and C/C/ND/NI. However, the hardest versions of the problem, including problem G/G/G/G, are actually NP-hard in the weak sense. Consequently, pseudo-polynomial time algorithms were developed to solve these problems. Recently, fast pseudo-polynomial algorithms for the most general problem (G/G/G/G) were proposed by van den Heuvel and Wagelmans (2006) and Chen et al. (2008).

Mathematical programming formulations of the uncapacitated SILSP can rather easily be extended to the capacitated case. For example, in the aggregate formulation (AGG) (see Section 2.2), it is enough to replace constraints (6) with $X_t \leq \min\{C_t, d_{tT}\}Y_t$ ($\forall t$). More general forms of capacity constraints were also studied. Chubanov et al. (2008) propose a model in which variables X_t are interpreted as the amount of a resource used for production. Then they introduce a function $g_t(X_t)$ which corresponds to the total number of manufactured units of the product given that X_t units of the resource were used. This model allows the formulation of different SILSP extensions with nonuniform resource capacities or resources with imperfect yield. Extended versions of these formulations were developed and were successfully solved by MIP solvers (see for example Pochet and Wolsey (2006)). Another application of these formulations is the Lagrangian heuristic presented by Brahimi and Dauzère-Pérès (2015) which exploits the mathematical program to generate better bounds.

Other exact solution approaches for the capacitated SILSP include branch-and-cut algorithms and the development of strong formulations to be solved by a MILP solver (Akbalik and Penz (2009), Akbalik and Pochet (2009), Atamtürk and Muñoz (2004)). Valid inequalities are usually derived by analyzing continuous 0-1 knapsack problems. It is also worth noting the combined dynamic-programming/B&B algorithm proposed by Chung et al. (1994).

In recent studies, Chubanov et al. (2006), Chubanov et al. (2008) and Ng et al. (2010) propose FPTAS for the capacitated SILSP with backlogging and with monotone cost structure. The running times of both procedures are not very relevant in practice. Finally, Chubanov and Pesch (2012) develop an FPTAS for a capacitated SILSP where demands can take negative values in which case they correspond to supplies.

Another similar research stream, which seems to be less explored, is the consideration of approximation formulations such as the ones proposed by Bitran and Matsuo (1986), Coleman and McKnew (1995), and Hardin et al. (2007). The latter focuses on the G/Z/Z/G problem. These formulations are supposed to be easier to solve and give acceptable error bounds or even optimal solutions in many cases (see Coleman and McKnew (1995)).

6.2. Inventory constraints

This section focuses on problems with special inventory characteristics, in particular problems with bounded storage capacity and/or perishable inventory.

6.2.1. Inventory capacity

Love (1973) was the first author to address the uncapacitated SILSP with inventory bounds. Modeling this problem is straightforward; a lower bound and an upper bound are set for inventory variables. Love (1973) showed that the problem can be solved using an $O(T^3)$ dynamic programming algorithm by decomposing the problem into regeneration intervals. Gutiérrez et al. (2001) propose a dynamic programming algorithm with the same complexity $(O(T^3))$ but it runs faster than the algorithm of Love (1973). This complexity is reduced to $O(T^2)$ in Toczylowski (1995), Liu (2008). Önal et al. (2012) showed that Liu's $O(T^2)$ algorithm was not correct. They proposed a fix to Liu's algorithm which runs in $O(T^2)$ time. When all costs are linear and production variables are integer, this complexity is reduced to $O(T \log T)$.

Atamtürk and Küçükyavuz (2005) identify facet defining inequalities for the uncapacitated SILSP with bounded inventory. They considered two models: A first model with linear cost on inventory and a second model with linear and fixed costs on inventory. Atamtürk and Küçükyavuz (2008) develop an $O(T^2)$ dynamic programming algorithm to solve the uncapacitated SILSP with inventory bounds and fixed costs on inventory. van Vyve and Ortega (2004) study the uncapacitated SILSP with fixed costs on inventory. They present an $O(T \log T)$ dynamic programming algorithm and the convex hull of integer solutions. Gade and Küçükyavuz (2011) complete these results and exhibit several challenges.

Several authors considered inventory bounds with other characteristics such as: backlogging (Hwang and van den Heuvel (2012)), lost sales (Hwang et al. (2013), Liu and Tu (2008), Liu et al. (2007), Loparic et al. (2001)), outsourcing (Chu and Chu (2007), Chu et al. (2013)), capacity constraints (Erenguc and Aksoy (1990), Akbalik et al. (2015)), delivery time windows (Jaruphongsa et al. (2004)). In a recent study, van den Heuvel and Wagelmans (2008) showed that the following lot-sizing problems are equivalent to the lot-sizing problem with inventory bounds: the lot-sizing problem with a remanufacturing option, the lot-sizing problem with production time windows, and the lot-sizing problem with cumulative capacities. Absi and Kedad-Sidhoum (2009) studied safety stocks which is a soft version of lower bounds on inventories. The lower bound on stock is a target level rather than a strong constraint. The authors developed a polynomial time dynamic programming algorithm for the uncapacitated single-item version.

A survey of heuristics applied to dynamic demand lot-sizing with limited warehouse capacity is presented in Minner (2009).

6.2.2. Perishable inventory

The uncapacitated SILSP with perishable inventory considers a deterioration rate for the product in stock. Inventory holding costs depend on how long a product remains in the inventory. This is, for instance, the case for food, pharmaceuticals, chemicals and blood.

Several papers consider inventory deterioration and perishability in continuous time lot-sizing models (e.g. Ghare and Shrader (1963)). This subject was studied for discrete time models by Friedman and Hoch (1978) and Rajagopalan (1992). Nahmias (1982) extensively discusses the issue of perishability. Hsu (2000) develops a dynamic programming algorithm to solve the uncapacitated SILSP with perishable inventory. The algorithm is based on a graph representation of the problem and runs in $O(T^4)$. Readers interested in this topic can refer to Pahl and Voß (2014) who propose a recent state-of-the-art review on production and supply chain planning models that consider deterioration and lifetime constraints.

Recently, Önal et al. (2015) consider the uncapacitated and capacitated SILSP where each item has a deterministic expiration date. They study four mechanisms to allocate items to the consumers. They show that the problem can be solved in polynomial time with all allocation mechanisms in the uncapacitated case, and become NP-hard for two allocation mechanisms in the capacitated case with time-invariant capacity. Dynamic programming algorithms, of complexity $O(T^2)$, $O(T^3)$ or $O(T^4)$ depending on the assumptions, are proposed for the polynomial problems. Finally, the two level lot sizing problem with perishable items was studied in Önal (2016). He shows that determining optimal procurement (at the first level) and transfer (at the second level) plan is NP-hard, and presents polynomial algorithms for special cases.

6.3. Constraints on lot sizes

6.3.1. Minimum order quantities

There are at least two situations where problems with Minimum Order Quantity (MOQ) are relevant. The first situation is when a MOQ is imposed by the supplier or by some technical constraints (see for example Lee (2004)). The second situation, noticed by researchers interacting with companies, occurs when production managers prefer to impose a MOQ restriction instead of estimating setup costs (Okhrin and Richter (2011)). Hence, a MOQ is an alternative approach for reaching economies of scale. Recently, Absi et al. (2016) showed that the uncapacitated SILSP with MOQ is NP-Hard.

Anderson and Cheah (1993) study a multi-item lot-sizing problem with MOQ and setup times. They developed a Lagrangian relaxation heuristic which decomposes the problem into a set of subproblems including single-item lot-sizing problems with minimum batch sizes. They propose a forward dynamic programming algorithm to solve these single-item problems. The authors do not specify the time complexity of their algorithm which might be exponential in the worst case. Special cases of the SILSP with minimum lot size were solved by Okhrin and Richter (2011) and Okhrin and Richter (2011). Okhrin and Richter (2011) considered the uncapacitated single-item lot-sizing problem with time independent MOQ for which they developed an $O(T^2)$. Okhrin and Richter (2011) propose an $O(T^3)$ algorithm to solve a special case of the problem where upper and lower bounds on production levels in addition to unit production and holding costs are constant. Hellion et al. (2012) extended this work by considering a problem with concave production and holding costs. They propose an exact algorithm to solve this problem in $O(T^6)$. Recently, Park and Klabjan (2015) studied the polyhedral structure of some SILSPs with constant MOQs and with constant and infinite production capacity. They also propose a polynomial time algorithm for the the SILSP with constant capacities and timeindependent MOQs. Park and Klabjan (2015) can also be considered as an excellent recent literature review of lot-sizing problems with production bounds.

6.3.2. Lot sizing with constant batch sizes or step-wise production costs

In the single-item lot-sizing problem with constant batch sizes or step-wise production costs (SICLSP-SW), production is made in constant batches of size v_t (which correspond to a vehicle size, for example) and there is a fixed cost (f_t) per batch. The number of batches is an integer decision variable denoted by Z_t . Thus, the costs are step-wise, i.e. piece-wise linear with discontinuous steps. C_t denotes the production capacity at period t.

Minimize
$$\sum_{t=1}^{T} (s_t Y_t + p_t X_t + f_t Z_t + h_t I_t)$$
(12)

Subject to:(5) and

$$X_t \le C_t \times Y_t \qquad \qquad \forall t \qquad (13)$$

$$X_t \le v_t Z_t \tag{14}$$

 $Y_t \le Z_t \qquad \qquad \forall t \tag{15}$

$$Z_t \le \lceil \frac{\upsilon_t}{\upsilon_t} \rceil Y_t \tag{16}$$

$$I_0 = I_T = 0 \qquad \qquad \forall t \tag{17}$$

$$Y_t \in \{0, 1\} \qquad \qquad \forall t \qquad (18)$$

$$Z_t \ge 0$$
 and integer $\forall t$ (19)

$$I_t, X_t \ge 0 \qquad \qquad \forall t \tag{20}$$

The objective function (12) minimizes the classical production and inventory costs plus the cost related to the number of batches. Constraints (5) and (13) are respectively the classical inventory balance equations and capacity constraints. Constraints (14) relate production variables and batch variables while Constraints (15) and (16) relate setup variables and batch variables. Constraints (17)-(20) define the decision variables. van Vyve (2007) considers the SICLSP-SW without setup costs. He develops polynomial time algorithms for cases with and without backlogging. Akbalik and Pochet (2009) propose a new class of valid inequalities called mixed flow cover inequalities and developed cutting plane algorithms. Akbalik and Rapine (2012) consider the SICLSP-SW with constant capacities and developed a polynomial time algorithm that runs in $O(T^4)$ for the case where production capacity is a multiple of the batch size and an $O(T^6)$ -time algorithm for the case with arbitrary time-independent capacity.

Lot-sizing problems with constant batch sizes are directly related to problems with more general cost structures (ex. problems with non-linear production costs). The link between the two is detailed in Section 9.3. A similar problem is the integrated lot-sizing and vehicle dispatching problem studied by Lippman (1969) and Alp et al. (2003), for example, and for which more details are presented in Section 8.2.

The more general case where batch size can vary over time is studied by Akbalik and Rapine (2013). They propose a classification of uncapacitated SILSP with time-dependent batch sizes to identify NP-hard and different polynomial cases of the problem.

6.4. Subcontracting and/or outsourcing

It seems that the borderline between subcontracting and outsourcing is not very clear in the Operations Management literature. In lot sizing, both terms were used interchangeably. It is still important to recall a definition which is commonly used in Operations Management textbooks such as Stevenson (2014) where *subcontracting* enables the company to acquire temporary capacity while *outsourcing* is a contract with another company to provide some goods or services on a regular basis.

Chu et al. (2013) consider the uncapacitated SILSP with outsourcing/subcontracting, backlogging and limited inventory capacity. The backlogging level at each period is supposed to be limited. The authors show that this problem can be solved in $O(T^4 \log T)$. Huang et al. (2008) introduce a polynomial time algorithm for a SILSP with outsourcing, backlogging, and non-decreasing inventory capacity. Wang et al. (2011) propose a dynamic programming algorithm that runs in $O(T^2)$ to solve a SILSP with remanufacturing and outsourcing without backlogging.

7. Complex structures

7.1. Multiple levels

In production planning with multi-stage, multi-level or multi-echelon systems, the product can be produced and/or stored at different stages. Transportation costs and times from one stage to another can be considered or ignored. These problems can occur in different supply chain structures: Serial systems, single-source multi-destination systems, multi-sourcing systems, or more general structures.

Serial supply chain systems occur when value is added to a product in a sequence of production facilities. Zangwill (1969) propose a dynamic programming algorithm that runs in $O(LT^4)$ to solve a serial multi-stage problem with T time periods and L levels (L > 2) and without capacity constraints. The same algorithm runs in $O(T^3)$ when L = 2 and production capacities are constant. Kaminsky and Simchi-Levi (2003) work on a three-level model in which the first and third levels are production stages, and the second level is a transportation stage. While production stages are capacitated, the transportation stage is supposed to have an infinite capacity. In Lee et al. (2003), several structural properties and a polynomial time algorithm is presented for a two-level problem with a stepwise transportation cost function. van Hoesel et al. (2005) propose a polynomial time $(O(T^7))$ dynamic programming algorithm to solve a two-level and multi-level problems with constant capacities. Melo and Wolsey (2010) solve an uncapacitated two-level lot-sizing problem in $O(T^2 \log T)$ and propose an extended formulation with $O(T^3$ variables and $O(T^2)$ equality constraints. Zhang et al. (2012) consider the multi-echelon lot-sizing problem in series without capacity constraints where the output of an intermediate echelon has also its own external demand. For the version with two levels, they propose an $O(T^4)$ dynamic programming algorithm to solve the problem optimally and gave a tight compact extended formulation. A hierarchy between the alternative formulations is also established. Denizel et al. (2010) consider the two-level and three-level uncapacitated serial lot-sizing problems. They propose a strong formulation based on the shortest path representation, together with $O(T^3)$ and $O(T^4)$ dynamic programming algorithms to solve the two-level and three-level lot-sizing problems, respectively.

In a single-source multi-destination system, there is one single facility which distributes the product to a set of destination facilities to add value to the product or to sell it to the consumer. The basic problem is mostly known as the one-warehouse multi-retailer (OWMR) problem. In the OWMR problem, there is a single warehouse which orders from its supplier to replenish a set of retailers. Each retailer faces its own external demands. The products can be inventoried at the warehouse or at the retailers. Replenishing the retailers from the warehouse and replenishing the warehouse from the supplier incur fixed ordering costs. Other costs include the per-unit purchasing and inventory holding costs. The OWMR problem consists of determining the quantities to be ordered in each time period from the warehouse and from the retailers. The OWMR problem is an extension of the Joint Replenishment Problem (JRP) and it is NP-hard (Arkin et al. (1989)). In the latter, inventory cannot be kept in the warehouse. This relationship was used by Arkin et al. (1989) to show that the OWMR problem is NP-hard. Recent reviews of the rich literature on the JRP can be found in Khouja and Goyal (2008) for the static demand models and in Robinson et al. (2009) for the dynamic demand case, where the name coordinated lot-sizing problem is used instead of joint replenishment problem. A recent literature review of the OWMR problem can be found in Solyali and Süral (2012), who present strong formulations and classifications for variants of the OWMR problem with dynamic demands. A special problem close to both OWMR and JRP is the one studied by Anily and Tzur (2005).

The multi-sourcing lot-sizing problem models several real-life situations such as multiple parallel machines, multiple transportation modes, or supplier selection (Aissaoui et al. (2007)). The multi-sourcing uncapacitated SILSP (MS-SILSP) is a direct generalization of the uncapacitated SILSP. It is defined by a given number of sources M from which we can order the single product. With each source m is associated a time dependent setup cost f_{mt} and a time dependent unitary production cost p_{mt} . The storage is done in a unique depot. Production variables (x_{mt}) and setup variables (y_{mt}) are indexed by m and t. The goal is to minimize the total production and setup costs as well as inventory costs. The flow conservation constraints and production constraints are modeled as follows:

$$s_{t-1} + \sum_{m=1}^{M} x_{mt} = d_t + s_t \ \forall t \in \{1, \dots, T\}$$
(21)

$$x_{mt} \leq \sum_{t'=t}^{T} d_{t'} y_{mt} \ \forall t \in \{1, \dots, T\} \ \forall m \in \{1, \dots, M\}$$

$$(22)$$

To the best of our knowledge, the MS-SISLP was not addressed in the literature most probably

because of its simplicity. Solving this problem is a direct generalization of the improved version of the Wagner and Whitin algorithm. It can be solved with an $O(MT \log T)$ dynamic programming algorithm. Several studies addressed more general versions of the MS-SISLP. Akbalik and Penz (2009) addressed the MS-SISLP with constant capacities and integer production variables. They solved the problem optimally using a pseudo-polynomial dynamic programming algorithm. Recently, Absi et al. (2013) addressed the multi-sourcing lot-sizing problem with different carbon emission constraints. In a case study, de Toledo and Shiguemoto (2005) consider a lot-sizing problem of a single item in several production centers without capacity constraints.

A more general supply chain structure was addressed by Yilmaz and Çatay (2006) who consider a single-item, multi-supplier, multi-producer, and multi-distributor production/distribution network. They propose relaxation-based heuristics to solve the problem. Finally, in a case study paper, Brahimi and Khan (2014) introduce a MIP formulation, solved using a commercial solver, for an extended single-item three-stage problem in the lube oil industry. The problem also integrates warehouse location decisions and allows different vehicles with different capacities to be used.

7.2. Remanufacturing

The last decades have seen the birth of several concepts associated with the development of green supply chains or sustainable supply chains. One of the most common concept is *reverse logistics*, which includes all processes that support the return of used products for recycling or reuse, such that disassembly, remanufacturing and refurbishing. Remanufacturing is the set of disassembly and recovery operations to repair a product so that it is equivalent to the original product. Generally, a remanufactured product must meet the same client expectations as new products. Guide et al. (1999) presented the general scheme of remanufacturing systems including remanufacturing or reuse of components from the disassembly of returned products.

The uncapacitated SILSP with a remanufacturing option (SILSP-R) is defined with known quantities of returned products in each period. These products can be remanufactured and considered as new ones. Customer demand can be satisfied from two sources (manufactured and remanufactured products). The goal is to determine when production takes place and when remanufacturing takes place and how many products are manufactured and how many are remanufactured. The objective is to minimize the classical total cost plus remanufacturing costs and holding costs for returned products. Teunter et al. (2006) consider two variants of the uncapacitated SILSP with a remanufacturing option. In the first one, manufacturing and remanufacturing have separate setup cost (SILSP-Rs). In the second one, manufacturing and remanufacturing have a joint setup cost (SILSP-Rj). Teunter et al. (2006) introduce an aggregate MIP formulation of SILSP-Rs. In addition to customer demand d_t , additional parameters are required for each period t: the number of returns, the unit holding costs for serviceables (product delivered to customers) and returns, the setup costs for manufacturing and remanufacturing, and the unit production costs for manufacturing and remanufacturing. Moreover, it is necessary to differentiate the inventory of serviceables from the inventory of returns, the number of manufactured items from the number of remanufactured items, the manufacturing setup from the remanufacturing setup. For the problem variant with joint setups (SILSP-Rj), the MIP model can be slightly modified.

Richter and Sombrutzki (2000) were among the first researchers who considered remanufacturing in the uncapacitated SILSP with stationary costs. They made the strong assumption that the number

ACCEPTED MANUSCRIPT

of returned products is sufficient to satisfy all demands. They solved the problem using a Wagner and Whitin like algorithm. Richter and Weber (2001) enriched the previous model by considering variable manufacturing and remanufacturing costs. Golany et al. (2001) studied the same problem in which it is possible to dispose returned products. They showed that the problem is NP-hard for general concave costs and solved it in $O(T^3)$ when all costs are linear. Yang et al. (2005) showed that the same problem is NP-hard even with time-invariant costs. Teunter et al. (2006) showed that the SILSP-Rj with stationary costs can be solved with an $O(T^4)$ dynamic programming algorithm. Recently, Fazle Baki et al. (2014) showed that the SILSP-Rs is NP-Hard. Retel Helmrich et al. (2014) showed that the general SILSP-Rj is NP-Hard and that the SILSP-Rs is NP-Hard even if all costs are stationary.

Teunter et al. (2006) propose modifications of classical heuristics (Silver-Meal, Least Unit Cost, and Part Period Balancing) to solve the SILSP-R. Recently, Schulz (2011) proposes an improvement of the modified Silver-Meal heuristic for the SILSP-Rs.

Pan et al. (2009) studied special cases of the capacitated lot-sizing problem with production, disposal, and remanufacturing. With only disposal or remanufacturing the problem can be converted into a capacitated lot-sizing problem. When disposal and remanufacturing capacities are considered, they propose a pseudo-polynomial time algorithm. For the uncapacitated production and capacitated remanufacturing case, they propose an exponential time algorithm. Zhang et al. (2012) deal with a capacitated lot-sizing problem in which the demands for manufactured and remanufactured products are distinct, share the same production resources but have different setup costs. They use a Lagrangian relaxation approach of capacity constraints to solve the problem. In Parsopoulos et al. (2015), the performance of a Differential Evolution (DE) algorithm is investigated.

8. Integrating other decisions

It has been shown in many recent studies that coordinated models integrating production planning with other types of decisions in companies (e.g. distribution, scheduling, etc.) generates higher profits or reduces costs. A survey of integrated models with single-item lot-sizing problems is presented below.

8.1. Scheduling

The goal in scheduling problems is to assign and sequence jobs (or tasks) on one or multiple machines to minimize operational objectives such as the maximum completion time of all jobs (makespan) or the sum of the delays (compared to given due dates) of the jobs. In a production setting, jobs are often lots whose quantities are decided by solving lot-sizing problems. Most integrated lot-sizing and scheduling problems are multi-item problems since the complexity of integrating scheduling decisions is often due either to the setup times between two consecutive lots of different items (as in the Discrete Lot-sizing and Scheduling Problem, DLSP, see Fleischmann (1990)) or to the series of operations (route) for each lot that must be sequenced on multiple machines (as in the job-shop lot-sizing and scheduling problem, see Lasserre (1992)).

To our knowledge, the only papers that consider single-item integrated lot-sizing and scheduling problems, more precisely the single-item DSLP, are the ones of Gavish and Johnson (1990), van Hoesel et al. (1994) and van Eijl and van Hoesel (1997). The single-item lot-sizing problems surveyed in the previous sections are usually tactical problems that consider the production planning of one product in relatively long time periods (e.g. days or weeks), while the DLSP considers micro-periods (e.g.

hours). Each time period usually has three states: Fully used for a setup, fully used for a production, or unused (off). Although the problem is not called single-item DLSP, Gavish and Johnson (1990) propose an approximation scheme, based on a dynamic programming algorithm, that converges to solution that is less than or equal to ϵ units of the optimal objective function. van Hoesel et al. (1994) introduce two approaches to solve the problem. The first approach is based on a strong linear programming formulation whose solutions are integer for some conditions. The second approach is a dynamic programming algorithm that runs in $O(T + D \log D)$ where D is the cumulative demand on the horizon. For the problem without speculative motives, van Eijl and van Hoesel (1997) present a partial linear description of the convex hull of feasible solutions.

8.2. Warehouse location, transportation and vehicle routing

The problem resulting from integration of lot-sizing decisions with transportation and vehicle routing is usually called the production routing problem (PRP). This problem can be considered as the combination of a two-level lot-sizing problem (LSP) and the vehicle routing problem (VRP). The two-level LSP considers the production and shipping of the demand for one or more items over a planning horizon of T time periods to a set of destinations (e.g. retailers, see Section 7.1). The VRP consists of determining the best routes to follow to deliver the planned quantities or demands to destinations.

The PRP has been significantly studied in the last decade. Adulyasak et al. (2015) present the most recent and comprehensive survey of research conducted on the PRP. Out of the 17 algorithms presented in this survey, 14 were published after 2006. Most of the research on the PRP concerns singleitem problems with limited plant production capacity (e.g. Boudia et al. (2007), Bard and Nananukul (2010)). Problems without capacity constraints were solved in Archetti et al. (2007), Ruokokoski et al. (2010), and Archetti et al. (2011) using branch-and-cut algorithms; in Absi et al. (2015) using an iterative heuristic; and in Adulyasak et al. (2012) using a large neighborhood search heuristic. Boundedness of inventory is sometimes considered at both the plant and the destinations or only at the destinations (e.g. Archetti et al. (2011)). In terms of distribution, studies have considered a single vehicle and multiple vehicles in addition to limited and unlimited fleet size. It seems that the only work that considers a heterogeneous fleet of vehicles is the paper of Lei et al. (2006). Cost components that are commonly considered with respect to production planning decisions are production fixed and variable costs and inventory holding costs either at the plant, at the destinations or at both levels. The problem with variable production costs is considered in Bertazzi et al. (2005) and Shiguemoto and Armentano (2010). For the distribution part, the classical traveling cost between points of sales is calculated based on the traveled distance. A single-item PRP with stochastic demands is solved by Adulyasak et al. (2015) using exact solution approaches based on Benders decomposition Benders (1962).

8.3. Order acceptance and market selection

An important decision in a company is order acceptance which is related to the process of receiving orders, negotiating due dates, and accepting or rejecting the orders. Order acceptance decisions are usually made in the sales department of a company. It seems that Geunes et al. (2002) are the first to consider an integrated production planning/order acceptance problem. Production planning models without setup costs and with order acceptance were considered by Aouam and Brahimi (2013) and Brahimi et al. (2015). In the former, a robust optimization approach was used to solve a problem

where orders can be partially accepted, while the latter proposes relax-and-fix heuristics to solve a deterministic problem where an order is either fully accepted or fully rejected.

In market selection problems, the decision maker has to choose which markets to satisfy over a whole production planning horizon. Thus, production is scheduled to satisfy the entire set of orders of selected markets, while no orders will be satisfied from "rejected" markets. An early study on the integration of lot-sizing decisions with market selection can be found in Levi et al. (2005), in which a polynomial-time approximation algorithm with constant-factor is proposed for a single-item lot-sizing problem with market selection and facility location decisions. van den Heuvel et al. (2012) consider the uncapacitated SILSP with market selection. They showed that the problem is strongly NP-hard by reduction from the 3SAT problem. While the classical uncapacitated SILSP is the most known polynomially solvable case of the market selection problem, there are other polynomially solvable cases which were presented and solved in van den Heuvel et al. (2012). These cases include problems with seasonal demands and problems with market-specific prices, among others. For example, using dynamic programming the former problem can be solved in $O(M(\log M + T \log T))$, where M is the number of markets.

8.4. Supplier selection

The objective of supplier selection decisions is to minimize purchasing and inbound logistics costs while maximizing delivery performance. When supplier selection decisions are integrated with lot sizing decisions, it is possible, for example, to explicitly consider the transportation component in the setup/ordering cost which will have a direct impact on lot sizes. Choudhary and Shankar (2014) addressed a problem in which a buyer procures a single product in multiple periods from multiple suppliers. They propose a multi-objective integer linear programming model for joint decision making of inventory lot-sizing, supplier selection and carrier selection problem. Moqri et al. (2011) studied a similar problem which they solved using a forward dynamic programming algorithm. For a comprehensive survey on integrated supplier selection and production planning models, see Aissaoui et al. (2007). For a general introduction to the topic of supplier selection we refer the reader to Ghodsypour and O'Brien (1998).

It is also important to see the similarity between some supplier selection problems and multisourcing lot-sizing problems surveyed in Section 7.1.

8.5. Supply chain coordination

Egri et al. (2014) studied a coordination problem between supplier and retailer. They assumed that the two parties do not have complete information about each other. The retailer knows that the supplier is solving (optimally) an uncapacitated SILSP with backlogging. The retailer aims at eliciting the cost parameters of the supplier's decision problem based on a historic record of (demand, delivery lot-size) pair. An inverse combinatorial optimization approach is used for this purpose. Hellion et al. (2014) studied the capacitated SILSP considering coordination between a company (e.g. a retailer) and its suppliers.

9. Other extensions

9.1. Stochastic models

In Section 5.4, a brief introduction was given on lot-sizing models with stochastic demands. This section presents a survey and classification of general stochastic lot-sizing studies, i.e. including studies

which consider costs, yield, lead time, or resources as stochastic parameters.

Silver (1978) stressed the importance of carrying out studies on lot-sizing problems in uncertain environments. Wemmerlöv (1989) studied the impact of forecast errors on the performance of different lot-sizing procedures including the dynamic programming algorithm of Wagner and Whitin (1958) and the heuristic of Silver and Meal (1973). One of the conclusions drawn by Wemmerlöv (1989) is that forecast errors lead both to stockouts and larger inventories.

Among earlier surveys on stochastic lot-sizing problems are Yano and Lee (1995) on lot sizing with random yields in continuous models, Koh et al. (2002) on uncertainty in an MRP environment, Gupta and Maranas (2003) on uncertainty in a supply chain with multiple products, and Mula et al. (2006). Tempelmeier (2013) proposes a survey of dynamic lot-sizing problems with demands as the only random parameters. His survey is particularly relevant to this section as he mainly focuses on single-item lot-sizing problems. We mostly relied on Aloulou et al. (2014), that recently present a quite comprehensive bibliography of the research on stochastic problems since 2000, to build this section of our survey. Aloulou et al. (2014) propose a classification based on five components A, B, C, D, and E which correspond to the number of periods, number of products, number of machines, uncertain parameter and modeling approach, respectively.

Based on the classification of Aloulou et al. (2014), we focus on papers classified as (T,1,1,D,E) and (T,1,m,D,E), i.e. Problems with multiple periods, single item, and single or multiple machines. Field D of the notation represents the uncertain parameters which can be the demand, lead time, yield, production and setup times, production and inventory capacities, costs, and other resources. Field E represents the modeling approaches which include on-line decisions, game theory, probabilistic approaches, queueing theory, fuzzy logic, scenario-based approaches, interval arithmetic, and simulation models.

Table 2 classifies stochastic SILSPs based on the stochastic parameters (columns) and the modeling approach (rows). It considers problems with single and multiple machines. Actually there are very few studies considering the stochastic SILSP with multiple machines. In multi-machine models (T,1,m,D,E), the same product is generated by different resources, which mostly correspond to different suppliers (Gebennini et al. (2009), Perakis and Zaretsky (2008), Sodhi (2005), Topaloglu (2005), Wang (2009), Yan and Tang (2009)). All of these references consider demand as the random parameter, except Wang (2009) where cost is the random parameter. While most studies are based on probabilistic models, studies in Sodhi (2005) and Wang (2009) are based on scenario formulations and fuzzy logic, respectively.

Some studies combine different stochastic parameters. For example, Huang and Ahmed (2010), and Senyiğit et al. (2013) consider that both demand and cost are uncertain, while the random parameters are demand cost, and capacity for Guan and Liu (2010), and Guan (2011).

The stochastic lot-sizing problem can be modeled with different approaches that use different optimization techniques. Generally, the main goal is to ensure a given service level. It is modeled using chance constraints, which means that the probability of reaching a given service level is larger than or equal to a given value. This service level can be defined according to the objective of the model. For example, some models can allow backlogging while others consider lost sales. Tempelmeier (2013) gave an overview of some existing service levels (α, β, δ and γ service levels). To deal with uncertainties, different strategies were proposed in the literature (see Bookbinder and Tan (1988), Tempelmeier (2013) and Küçükyavuz (2011)).

	Demand	Cost	Yield	Lead Time	Capa.
Probabilistic	[45], [65], [69], [72], [87],	[237], [141],	[271],	[22]	
	[85], [86], [175], [209], [212],	[225]	[221],		
	[233], [238], [239], [241], [243],		[130],		
	[242], [250], [267], [266]		[103]		
Scenario-based	[121], [120], [128], [133], [240],	[124], [122],	[53]	[151]	
	[289]	[128], [292]			
On-line decisions	[1], [270]				
Fuzzy logic	[126], [156]	[156]	[274]		[203]
Interval arithmetic	[153]				
Simulation	[213], [101]				

Table 2: Stochastic lot sizing: Uncertain parameters and modeling approaches

- Static strategy: In this strategy, all production decisions (setup periods and produced quantities) are taken in advance and the whole production plan is executed independently of the actual realization of demands. This model is particularly useful when capacity is limited and a capacity-feasible production plan must be defined in advance.
- **Dynamic strategy:** Here the decision is made for only one period based on forecasts of the whole planning horizon. Once the demand of the current period is revealed, a new production plan is established for the rest of the horizon. In fact, one may revise the production plan according to the outcome of observed values of random variables.
- Static-Dynamic strategy: This is a hybrid strategy in which all setup decisions are established for the whole horizon, but production quantities are updated at each period according to observed demands of previous periods. This means that, once the random variables are observed, one may revise only production quantities without changing setup decision variables.

In order to deal with MIP models, modeling approaches are generally based on a set of scenarios or a scenario tree. Guan and Miller (2008) considered the simplest version of the stochastic SILSP based on a scenario tree. They propose a polynomial time dynamic programming algorithm with respect to the number of scenarios and the number of periods. Guan et al. (2006) propose a branch-and-cut algorithm to deal with the same problem. Guan (2011) deals with the stochastic SILSP with backlogging, and introduces a polynomial dynamic programming with respect to the number of scenarios and the number of scenarios are problem. Guan (2011) deals with the stochastic SILSP with backlogging, and introduces a polynomial dynamic programming with respect to the number of scenarios and the number of periods. Küçükyavuz (2011) provides a relevant tutorial on the stochastic SILSP that groups her work and cite several relevant works.

Another approach to deal with lot-sizing problems with random demands is robust optimization (see Ben-Tal et al. (2009)), where feasible solutions are determined for a range of scenarios of the uncertain parameters. Contrary to other stochastic models, the probability distribution of the random parameter does not need to be known. There are very few studies considering robust optimization in production planning. To the best of our knowledge, the following papers are the only ones which use robust optimization (based on the approach of Ben-Tal et al. (2009)) to solve production planning problems: Bertsimas and Thiele (2006), Aouam and Brahimi (2013), and Brahimi et al. (2015).

Finally, Levi and Shi (2013) develop approximation algorithms for the stochastic SILSP with random demands, lead times and dynamic forecast updates. A randomized policy is proposed with a worst-case performance guarantee of 3.

9.2. Pricing

Most lot-sizing models assume that the price of the product is a exogenous parameter. However, it has been shown in different studies that pricing strategies and their integration with production/inventory decisions can considerably improve profitability. In this section, we define pricing as the process of changing product price (once or dynamically over time). Thus, price is a decision variable in pricing models. The discussion will be limited to those studies which consider demand as a function of the price and its value in each period has to be determined.

It seems that the earliest surveys on the topics of pricing and yield management are Gallego and van Ryzin (1994) and Kimes (1989), respectively. Later, Elmaghraby and Keskinocak (2003) and Chan et al. (2004) provide comprehensive reviews of the literature up to 2003/2004 and a practical introduction to the topic of pricing and its integration with production/inventory decisions. More recent surveys include Chen and Simchi-Levi (2012) and Simchi-Levi et al. (2014).

The above-mentioned surveys cover problems with single and multiple time periods, stochastic and deterministic models and single and multi commodity models, among others. In this section, we briefly survey papers integrating pricing decisions with dynamic demand SILSPs. We distinguish studies based on demand uncertainty, capacity limit, and sales considerations (backlogging/lost sales). Furthermore, the following characteristics specific to pricing literature will be considered: Price nature (static or dynamic), relationship between demand and price, and cost of price changing.

The earliest study that integrated dynamic lot sizing with pricing decisions is Thomas (1970), where a deterministic uncapacitated SILSP with concave costs and no backlogging is considered. With respect to *pricing nature*, Thomas' model is a *dynamic pricing model* in which prices are allowed to be changed at the beginning of each time period. Dynamic pricing was also considered in uncapacitated SILSPs with deterministic demand by Bhattacharjee and Ramesh (2000), Deng and Yano (2006) and Chen and Hu (2012). Lost sales are allowed in the first paper, while no backlogging or lost sales are allowed in the two other papers. Merzifonluoglu et al. (2007) study a deterministic model with subcontracting and overtime options.

Stochastic demand models with dynamic pricing can be found in Thomas (1974), Chen and Simchi-Levi (2004), and Chen et al. (2011) with infinite capacity and in Federgruen and Heching (1999) and Chen and Chen (2005) with limited capacity.

In static pricing models there is only one price change in the whole planning horizon. Kunreuther and Schrage (1973) consider a static pricing model with deterministic demands and where backlogging and lost sales are not allowed. Static pricing models were also presented in Gilbert (1999) and van den Heuvel and Wagelmans (2006). Using a numerical study Federgruen and Heching (1999) indicated that dynamic pricing may result in better profits over fixed pricing models. A particular model called model with delayed decisions was proposed by Chan et al. (2006). In this model, if production (resp. pricing) is determined at period t, changes in pricing (resp. production) will not take place until period t+k. A special case is when pricing is made at the beginning of the planning horizon and production decisions are made in the beginning of each time period. Different strategies were analyzed and the authors concluded, for example, that delayed production is usually better than delayed pricing.

The relationship between demand and price is represented by a function $D = \phi(P)$, where D is the demand and P is the price. An example of such a function is the linear relationship D = b - aP, where a and b are strictly positive real numbers. When demand is stochastic, it can be represented as a random variable whose distribution depends on the price. A general function assumed by Chen and Simchi-Levi (2004) is: $D_t(P_t, \epsilon_t) = \alpha_t D_t(P_t) + \beta_t$, where the random perturbation over time, ϵ_t is composed of the random variables α_t and β_t . There are two special cases of this model: The additive model and the multiplicative model. In the additive model, $\alpha_t = 1$ while $\beta = 0$ in the multiplicative model Chen and Simchi-Levi (2004). Chen and Simchi-Levi (2004) also present two special cases to function $D_t(P_t)$, both of which are common in the economics literature. The two cases are $D_t(P) = b_t - a_t P$ ($a_t > 0, b_t > 0$) and $D_t(P) = a_t P^{-b_t}$ ($a_t > 0, b_t > 1$), for the additive and multiplicative cases, respectively. Chan et al. (2006), considered a capacitated model with a discrete menu of prices; i.e. price in each time period must be chosen from a finite list. The relationship between price and expected demand is assumed to be known without necessarily having a specific shape. Federgruen and Heching (1999) study a stochastic demand problem with additive pricing. Geunes et al. (2006) consider a problem with piecewise-linear and concave revenue functions in price and finite production capacities.

Most pricing studies do not consider *price adjustment costs*. Chen and Hu (2012) study a deterministic uncapacitated SILSP in which a cost is incurred each time the price changes. This cost can be related to printing catalogs, for example. In another paper, Chen et al. (2011) extend the problem of Federgruen and Heching (1999) to include price changing cost. Furthermore, price changing cost in Chen et al. (2011) is composed of fixed and variable components.

In general, adding pricing considerations to SILSPs makes the problem harder to solve. However, easy special cases were identified and solved in polynomial time. The first polynomial time algorithm is a dynamic program proposed by Thomas (1970) to solve the deterministic uncapacitated SILSP. Thomas (1970) proves that there are several common properties between his integrated problem and the Wagner and Whitin (1958) model. He provides a forward dynamic programming algorithm to solve the problem. Kunreuther and Schrage (1973) propose a heuristic procedure, which provides upper and lower bounds on the price decision, for their static price model. Gilbert (1999) considers a problem similar to the one of Kunreuther and Schrage (1973) and develops an exact solution approach. Deng and Yano (2006) analyze properties of optimal solutions for constant and time-varying capacity, and with and without speculative motives. In particular, the authors show that the properties introduced in Florian and Klein (1971) for the capacitated SILSP can be extended. The remainder of the paper is devoted to providing managerial insights using the proposed solution procedures on various numerical examples.

The research on models with stochastic demands has been focused on identifying optimal ordering policies or developing heuristic algorithms. For the stochastic demands problem with backlogging, Thomas (1974) proposes a simple policy referred to as (s, S, P), where s is the reorder point, S is the order-up-to level of inventory, and P is the price. Although this policy is not optimal, Thomas (1974) states that it is optimal under fairly general conditions. Chen and Simchi-Levi (2004) prove that the (s, S, P) policy of Thomas (1974) is indeed optimal for the multiplicative model. They also propose an optimal policy for the general demand model. Chen et al. (2011) consider a stochastic demand uncapacitated SILSP, and for which they propose an intuitive heuristic procedure.

9.3. Cost structure

We limit the discussion to single-item lot-sizing problems where the objective is to minimize a function of the form:

$$\sum_{t=1}^{T} \left(f_t^p(x_t) + f_t^h(I_t) \right)$$



Figure 1: Examples of piecewise concave cost functions

where $f_t^p(.)$ and $f_t^h(.)$ are the production and the inventory holding cost functions, respectively. In the seminal paper by Wagner and Whitin (1958), an $O(T^2)$ algorithm was proposed for the problem where inventory holding cost is constant $(f_t^h(.)$ is a linear function) while the production cost function has the form: $f_t^p(x_t) = s_t \delta(x(t)) + p_t x_t$, whith $\delta(x(t)) = 1$ if $x_t > 0$ and zero otherwise. Both of these functions are concave. Wagner (1960) showed that the algorithm of Wagner and Whitin (1958) runs in $O(T^2)$ if f_t^p is concave and $f_t^h(I_t) = h_t I_t$. Later, Veinott, 1963 (as cited in Aggarwal and Park (1993)) showed that even if both f_t^p and f_t^h are arbitrary concave functions, the problem can be solved in $O(T^2)$.

Functions of the form $f_t^p(x_t) = s_t \delta(x(t)) + p_t x_t$ and $f_t^h(I_t) = h_t I_t$ are the most commonly used cost functions in the SILSP literature. The uncapacitated SILSP with no speculative motives is known to be solvable in O(T). Note that there are many practical instances having such cost structure (Wolsey (1995)).

Love (1973) considered general concave functions f_t^p and f_t^h in problems with bounded production and inventory capacities. Florian and Klein (1971) solved a capacitated SILSP where f_t^p and f_t^h are arbitrary concave functions. Atamtürk and Küçükyavuz (2008) solved a SILSP with bounded inventory and fixed charge inventory holding cost.

A generalization of the above functions is a piecewise concave function. If, for example, f_t^p is a piecewise concave function with breakpoints at $p^0 < p^1 < ... < p^m$, then f_t^p is concave in each of the m subintervals $[p^0, p^1], [p^1, p^2], ..., [p^{m-1}, p^m]$. For a background on piecewise concave functions and their link with lot-sizing problems, we invite the reader to check Zangwill (1967).

Swoveland (1975) solves a problem where f_t^p and f_t^h are piecewise concave. Backlogging is allowed and function f_t^h is defined as a holding-backordering cost function. That is $f_t^h(I_t)$ is equal to the positive inventory holding cost if $I_t \ge 0$ and is equal to the positive backordering cost of quantity $-I_t$ if $I_t \le 0$.

The advantage of piecewise concave functions, as proposed by Swoveland (1975) and Koca et al. (2014), is that it generalizes several lot-sizing problems including problems with quantity discounts, minimum order quantities, capacities, overloading and outsourcing. As it can be expected, there are uncapacitated SILSPs with some types of piecewise cost functions that were proven to be NP-hard (see for example, Chan et al. (2002)).

For the SILSP with convex costs, Kian et al. (2014) present a dynamic programming algorithm of complexity $O(T^22^T)$ in the general case, and $O(T^22^T)$ for zero fixed setup costs. Given the prohibitive complexity of the exact algorithm, six heuristics are also proposed to solve instances of reasonable sizes.

In the next paragraphs, we present some applications of generalized cost functions.

Modeling discounts. Federgruen and Lee (1990) study lot-sizing problems with both incremental and all unit discounts. Chan et al. (2002) study a problem with all unit discount where the cost function is represented in Figure 1(a). Li et al. (2012) consider the uncapacitated SILSP with all-unit discount with time invariant breakpoints of the cost function. They propose an $O(T^{m+3})$ algorithm to solve problems where the cost function has m breakpoints. Mirmohammadi et al. (2009) develop an exact B&B algorithm to solve an all unit discount problem. Hu et al. (2004) propose a modified silver and meal heuristic to solve the uncapacitated SILSP with incremental quantity discount. Choudhary and Shankar (2011) considered a problem with all unit discount, storage capacity, and a rejection rate. They propose an integer linear programming model and analyze the effect of variations in problem parameters such as rejection rate, demand, storage capacity and inventory holding cost. Archetti et al. (2014) studied an uncapacitated SILSP with cost discounts. They consider models with modified allunit discount cost functions and incremental discount cost functions, and develop polynomial time algorithms to solve some special cases.

Modeling constant batch size problems. In lot-sizing problems with constant batch size presented in Section 6.3, production costs are piecewise concave. For example, the production cost function $f_t^p(x_t)$ in Akbalik and Rapine (2012) is $s_t \delta(x(t)) + p_t x_t + p_t^b \lceil \frac{x_t}{B_t} \rceil$ (and $p_t x_t + p_t^b \lceil \frac{x_t}{B_t} \rceil$ in van Vyve (2007)), where p_t^b is a fixed cost per batch and B_t is the batch size (See Figure 1(b)).

Modeling outsourcing/subcontracting. In problems with subcontracting presented in Section 6.4, it is possible to represent the overall production/subcontracting cost by a concave function such as the one in Figure 1(c). This is the case of the problem studied in Atamtürk and Hochbaum (2001). A simplified version of the function in Figure 1(c) is when the two curves are linear functions, which is considered in Chu et al. (2013).

Modeling minimum order quantities. The production costs of lot-sizing problems with a minimum production quantity (see Section 6.3), such as those studied by Anderson and Cheah (1993) and Hellion et al. (2012), can be represented with a concave function such as Figure 1(d). Koca et al. (2014) present an overview of studies on lot-sizing problems with different piecewise concave functions, and propose a model that generalizes many single-item problems. A dynamic programming algorithm is developed to solve this problem and show that the algorithm has different complexities depending on the structure of the cost function. Hsu and Lowe (2001) studied an uncapacitated SILSP where

backorder and inventory holding costs are said to be "period-pair-dependent". This means that the holding cost (backordering) cost in a given period depends on the period where the order is produced (placed) and the period where it is used (filled). Since that the problem is NP-hard (Hsu and Lowe (2001)), they propose polynomial time dynamic programming algorithms to solve some special cases.

9.4. Other costs

Startup costs. Agra and Constantino (1999) studied an uncapacitated SILSP with setup and startup cost and backlogging. While the set-up cost is incurred when the machine is able to produce, the start-up cost is incurred whenever the machine is set up for the item and was not set up for that item in the previous period. After proving that the capacitated SILSP with startup costs is NP-hard, Hindi (1995) propose a tight formulation and a column generation algorithm. Wolsey (1989) and van Hoesel et al. (1994) propose different classes of valid inequalities for the uncapacitated SILSP with startup costs.

Escalante et al. (2011) consider the capacitated SILSP with *continuous startup costs* in which a continuous setup cost is incurred in period t if there a positive production in t and the production capacity in t - 1 was not saturated. They provide a polyhedral study of the problem.

Reservation costs. Toy and Berk (2006) and Berk et al. (2008) study a capacitated SILSP in a warm/cold process in which the process can be kept warm from period t to t + 1 at a reservation cost. Toy and Berk (2006) propose an $O(T^3)$ algorithm for the problem without shortages (full commitment), while Berk et al. (2008) introduce an $O(T^5)$ algorithm for a problem where lost sales are allowed. Berk et al. (2008) studied a capacitated SILSP in a warm/cold process in which the process can be kept warm from period t to t + 1 at a reservation cost. They propose an algorithm that runs in $O(T^5)$ time for a problem where lost sales are allowed.

Reservation and startup costs. Karmarkar et al. (1987) studied a capacitated problem with reservation and startup costs. Reservation cost is eventually used to keep the machine "warm" or "on" in a period where there is no production. In Coleman and McKnew (1995), a pure zero-one IP formulation was proposed for a special case of the capacitated problem with simplifying assumptions. The solution of the proposed model gives optimal solution for 93% of the tested problems without simplifying assumptions.

Load change cost. Zangwill (1966) studied the production smoothing problem in which a cost is incurred if production level changes from one period to another. He considered a problem with concave production cost, a concave inventory cost, and a piecewise concave cost of changing the production level from one period to the next. He developed dynamic programming algorithms for this problem and for the fixed charge case. The effect of lot-sizing on workload variability and cost of load change was studied by Askin (1983).

9.5. Environmental issues

Recently, several authors addressed lot-sizing with different environmental constraints. This includes remanufacturing (Teunter et al., Retel Helmrich et al. (2006, 2014)), carbon emission constraints (Absi et al., Absi et al., Retel Helmrich et al. (2013, 2016, 2015)) and co-production (Agrali (2012)). Carbon emissions constraints deal with several new legislative constraints that aim at reducing the overall environmental impact. These constraints were addressed with different point of views. Absi

ACCEPTED MANUSCRIPT

et al. (2013) considered carbon emission constraints that limit the unitary carbon emission following several concepts. They propose four types of carbon emission constraints: Periodic carbon emission constraint, Cumulative carbon emission constraint, Global carbon emission constraint, and Rolling carbon emission constraint. These constraints impose a maximum value not on the total carbon emission, but on the average carbon emission per product. This type of constraints is particularly relevant to the firms who want to display the carbon footprint of their products. Absi et al. (2013) propose a polynomial algorithm to solve the multisourcing single-item lot-sizing problem with periodic carbon emissions. They also show that the same problem with the three other types of constraints is NP-Hard. Absi et al. (2016) showed that the multisourcing single-item lot-sizing problem with periodic carbon emissions a fixed carbon emissions is NP-Hard and propose a pseudo-polynomial algorithm to solve it. Retel Helmrich et al. (2015) considered a global carbon emission constraint that limits the overall impact over the whole time horizon. They showed that the single-item lot-sizing problem with this global constraint is NP-Hard, they also propose a Lagrangian heuristic, pseudo-polynomial algorithms and a fully polynomial time approximation scheme (FPTAS).

10. Conclusions and discussions

In this paper, following our previous review Brahimi et al. (2006), we surveyed the literature on single-item lot-sizing problems and the important recent advances in this research area. The growing interest in lot-sizing problems, stemming both from their scientific significance and practical applications, led to the *International Workshop in Lot Sizing* (IWLS) which is held every year since 2010 and the creation of the EURO Working Group on Lot-Sizing LOT in 2014.

Even if single-item lot-sizing problems often do not directly model industrial problems, they generally correspond to subproblems of more complex industrial problems and are useful in decomposition approaches. They also help understanding some structural properties of these industrial problems. While concluding this literature review, we propose some research directions that we believe need more focus in the near future. These research directions mainly correspond to lot-sizing in sustainable supply chains, with stochastic parameters and integrated with other decisions.

A classification of the publications is presented in Table 3, which immediately shows some interesting research gaps to be filled. For instance, although we have shown in Section 8 studies integrating lot-sizing decisions with other types of decisions, such as scheduling, routing or market selection, new approaches for known problems could be investigated and new and relevant integrated problems should be explored. These problems are of importance since they show that research should cross topics and benefit from recent advances in multiple research areas. Recently, several papers addressed multiitem lot-sizing problems combined with cutting stock decisions. To the best of our knowledge, these problems are not addressed in their single-item version. Integrating the single-item lot-sizing problem with cutting stock leads to new original problems that can be helpful to solve the multi-item versions. Also, very few studies on the stochastic integrated maintenance and dynamic lot-sizing problem were conducted (See for example Kuhn (1997)). Due to the stochastic nature of breakdowns, this is an interesting and relevant problem to further explore. The coordination of decisions taken at different levels or by different actors (see Section 8.5) remain a challenging issue in SILSPs for which Game Theory is relevant.

The single-item lot-sizing problem with remanufacturing options was addressed in its deterministic version (see Section 7.2). Returns are supposed known in advance which is generally difficult to

ACCEPTED MANUSCRIPT

quantify unless the company manages all flows of returns. We believe that this problem should be addressed in its stochastic version where returns are stochastic parameters. The goal would be to plan production and remanufacturing when considering stochastic returns. Returns can also be remanufactured at different levels to produce final products, semi-finished goods or raw materials. If we go one step further, the concept of circular economy is now a hot topic in management science but remains relatively absent from the lot-sizing research. Circular economy aims at producing goods and services while minimizing the consumption and waste of raw materials and limiting the use of non-renewable energies. Circular economy includes the concept of reverse logistics by focusing on creating jobs and wealth through the choice of recycling levels. In the literature, the notions of by-product and co-product are not well defined. Both products are generated when producing the main product. A co-product has generally its own demand when a by-product is an undesired product. Agrali (2012) considered co-production in a multi-item environment where co-products have their own demand. Even though this problem is of importance, there is a lack of studies to deal with single-item lot-sizing problems with co-products or by-products. The circular economy concept goes beyond the environmental pillar since it considers also societal considerations. In the lot-sizing literature, this should encourage researchers to jointly consider carbon emissions constraints, remanufacturing/disassembly and by-products. The goal is also to model societal objectives. This will lead to the definition and the study of new original lot-sizing problems that can help answering questions related to circular economy.

Other interesting problems related to the control of the environmental impact that are becoming important in practice are lot-sizing problems with energy constraints or costs. Several recent surveys on energy-efficient scheduling problems are available in the literature (see for instance Gahm et al. (2016)) but, to our knowledge, nothing has been published in the lot-sizing literature so far.

The majority of studies addressing stochastic single-item lot-sizing problems deal with stochastic demands. In practice several parameters can be stochastic such as lead times, yields, production and setup times, prices, and availability of resources. Recent studies addressed some of these problems. For example, Huang and Küçükyavuz (2008) address the single-item lot-sizing problem with stochastic lead-times. They provide a dynamic programming algorithm in the scenario tree size to solve the studied problem. The main weakness of the stochastic lot-sizing models is that they are not widely used in industry. One of the main goals would be to show how these models can be useful for industry, e.g. how to define the relevant scenarios in order to obtain tractable models.

Acknowledgements

The authors would like to thank Safia Kedad-Sidhoum, Hark-Chin Hwang, Wilco van den Heuvel and Laurence Wolsey for their insightful comments on an earlier version of the paper.

Table 3: Summary and classification of publications. The corresponding number of references presented in Brahimi et al. (2006) and omitted from this updated review are identified with an asterisk (*).

Class/Subclass	Complexity	Dynamic Programming	Polyhedral/MIP/B&C	B&B and	Heuristics	Others		
			[40]	dual				
Basic Problem		$[2^{*}], [39], [30^{*}], [32^{*}], [67^{*}], [105^{*}], [269],$	[40]	[174],	$\begin{bmatrix} 36 \end{bmatrix}, \begin{bmatrix} 37 \end{bmatrix}, \begin{bmatrix} 44 \end{bmatrix}, \begin{bmatrix} 52 \end{bmatrix}, \begin{bmatrix} 20 \end{bmatrix}$			
		[109*]		[262]	[82], [88], [194], [229],			
Assumptions on Domand					[201], [200], [201], [210]			
Assumptions on Demand		[6] [75] [84] [120] [142] [145] [147]	[11] [166] [99*] [94*] [941]	[997]	[85] [86] [225]	[111] [264]		
Dacklogging		$ \begin{bmatrix} 0 \\ 0 \end{bmatrix}, \begin{bmatrix} 70 \\ 0 \end{bmatrix}, \begin{bmatrix} 64 \\ 0 \end{bmatrix}, \begin{bmatrix} 133 \\ 1 \end{bmatrix}, \begin{bmatrix} 143 \\ 1 \end{bmatrix}, \begin{bmatrix} 145 \\ 1 \end{bmatrix}, \begin{bmatrix} 147 \\ 1 \end{bmatrix}, \begin{bmatrix} 147 \\ 1 \end{bmatrix}, \begin{bmatrix} 127 \\ 1 \end{bmatrix}, \begin{bmatrix} 201 \\ 1 \end{bmatrix}, \begin{bmatrix} 233 \\ 1 \end{bmatrix}, \begin{bmatrix} 264 \\ 1 \end{bmatrix}, \begin{bmatrix} 288 \end{bmatrix} $	[11], [100], [02], [04], [241], [250], [263]	[221]	[65], [60], [225]	[111], [204]		
Lost Sales		$[6], [20], [46], [148], [180], [93^*]$	[73]		[49]			
Time Windows		[6], [83], [145], [144], [150], [168]			[59]	[254]		
Constraints on Resources	-							
Production Capacity	$[14], [11^*],$	$[14], [17], [46], [67], [14^*], [77], [79],$	$[16], [14], [35], [9^*], [61], [81],$	$[3^*], [18^*],$	[59], [77], [78], [79],	[132], [254]		
	[36*]	$[78], [17^*], [18^*], [102], [38^*], [162],$	$[91], [132], [60^*], [182], [75^*],$	[95], [132],	$[85], [86], [38^*], [49^*],$			
		$[63^*], [187], [193], [196], [201], [204],$	[210], [263], [284], [291]	[63*]	$[136], [193], [102^*],$			
		$[210], [94^*], [249], [253], [104^*], [101^*],$			[284], [291]			
		$[261], [102^*], [264], [286]$		[05]	[100]			
Inventory Constraints		$\begin{bmatrix} 15\\ 34\\ 75\\ 162\\ 162\\ 162\\ 162\\ 162\\ 164\\ 162\\ 164\\ 162\\ 164\\ 162\\ 164\\ 162\\ 164\\ 162\\ 164\\ 162\\ 164\\ 164\\ 164\\ 164\\ 164\\ 164\\ 164\\ 164$	[33], [91], [183], [282]	[95]	[188]	[254][202], [223]		
Device a bla Inconstanting		$[148], [150], [183], [04^{+}], [199], [247]$						
Constraint on Lot Sizes	[4] [19]	[37], [51], [198], [80]	[16] [205] [210]			[124] [105] [264]		
Constraint on Lot Sizes	[4], [10]	[17], [23], [190], [210], [204]	[10], [203], [210]			[154], [195], [204]		
Complex Structure		[10], [143], [187], [273]	[187]					
Multi ashalan	[90]	[00] [153] [508] [186] [361] [388] [300]	[61] [02] [994] [989] [984]		[994]	[02] [202]		
Multi comping	[29]	$[90], [152], [59^{\circ}], [160], [201], [260], [290]$	[01], [93], [224], [262], [264]		[284]	[95], [202]		
De menufe sturin r	[0], [14]	[0], [14], [10], [04]			[07%] [000] [044] [001]	[200] [27.4]		
Re-manufacturing	$[98], [40^{+}], [216], [281]$	$\begin{bmatrix} 985, [40^\circ], [175], [204], [87^\circ], [218], [244], \\ [273] \end{bmatrix}$	[98], [282], [291]		[87], [222], [244], [291]	[200], [254]		
Integrating Decisions								
Scheduling		[38*], [103*]	[103*]			$[27^*], [90^*]$		
Location, Transportation	[28]		[41], [61], [74], [284]		[2], [28], [55], [171]	[8]		
Market/Order Selection	[252]	[66]	[58]		[58], [252]			
Supplier Selection		[190]				[74]		
Supply Chain Integration			[93], [135]			[93]		
Other Extensions								
Stochastic Models	[120]	[120], [175], [212], [233]	$\begin{bmatrix} 25], & [42], & [58], & [121], & [122], \\ & [120], & [156], & [241], & [250] \end{bmatrix}$	[128]	$\begin{bmatrix} 65], & [69], & [86], & [85], \\ [87], & [225], & [243], & [246] \end{bmatrix}$	[8], [120], [156], [225], [238], [241]		
Pricing/ Discounts		[20], [68], [89], [118], [187], [176], [165], [187], [245], [255]	[187], [73], [187]	[189]	$\begin{bmatrix} [49], \ [65], \ [71], \ [140], \\ [246], \ [69] \end{bmatrix}$	[26]		
Cost Structure		$[17], [34], [46], [158], [162], [63^*], [193]$		[95], [63*]		[134], [223]		
Other Costs		[46], [249], [104*], [286]	[11], [81], [97], [259], [278]		[136], [49*]	[31]		
Co-production		[12]	[12]					
Others	[18], [139]	[6], [18], [46], [78], [139]		[128]		[111], [195], [283]		

References

References

- [1] Aarts, E. H. L., Reijnhoudt, M. F., Stehouwer, H. P., Wessels, J., 2000. A novel decomposition approach for on-line lot-sizing. European Journal of Operational Research 122 (2), 339–353.
- [2] Absi, N., Archetti, C., Dauzère-Pérès, S., Feillet, D., 2015. A two-phase iterative heuristic approach for the production routing problem. Transportation Science 49 (4), 784–795.
- [3] Absi, N., Dauzère-Pérès, S., Kedad-Sidhoum, S., Penz, B., Rapine, C., 2013. Lot sizing with carbon emission constraints. European Journal of Operational Research 227 (1), 55–61.
- [4] Absi, N., Dauzère-Pérès, S., Kedad-Sidhoum, S., Penz, B., Rapine, C., 2016. The single-item green lot-sizing problem with fixed carbon emissions. European Journal of Operational Research 248 (3), 849 – 855.
- [5] Absi, N., Kedad-Sidhoum, S., 2009. The multi-item capacitated lot-sizing problem with safety stocks and demand shortage costs. Computers & Operations Research 36 (11), 2926–2936.
- [6] Absi, N., Kedad-Sidhoum, S., Dauzère-Pérès, S., 2011. Uncapacitated lot-sizing problem with production time windows, early productions, backlogs and lost sales. International Journal of Production Research 49 (9), 2551–2566.
- [7] Adulyasak, Y., Cordeau, J.-F., Jans, R., 2012. Optimization-based adaptive large neighborhood search for the production routing problem. Transportation Science 48 (1), 20–45.
- [8] Adulyasak, Y., Cordeau, J.-F., Jans, R., 2015. Benders decomposition for production routing under demand uncertainty. Operations Research 63 (4), 851–867.
- [9] Adulyasak, Y., Cordeau, J.-F., Jans, R., 2015. The production routing problem: A review of formulations and solution algorithms. Computers & Operations Research 55 (0), 141 152.
- [10] Aggarwal, A., Park, J. K., 1993. Improved algorithms for economic lot size problems. Operations Research 41 (3), 549.
- [11] Agra, A., Constantino, M., 1999. Lotsizing with backlogging and start-ups: the case of Wagner-Whitin costs. Operations Research Letters 25 (2), 81–88.
- [12] Agrali, S., 2012. A dynamic uncapacitated lot-sizing problem with co-production. Optimization Letters 6 (6), 1051–1061.
- [13] Aissaoui, N., Haouari, M., Hassini, E., 2007. Supplier selection and order lot sizing modeling: A review. Computers & operations research 34 (12), 3516–3540.
- [14] Akbalik, A., Penz, B., Jun. 2009. Exact methods for single-item capacitated lot sizing problem with alternative machines and piece-wise linear production costs. International Journal of Production Economics 119 (2), 367–379.
- [15] Akbalik, A., Penz, B., Rapine, C., 2015. Capacitated lot sizing problems with inventory bounds. Annals of Operations Research 229 (1), 1–18.
- [16] Akbalik, A., Pochet, Y., 2009. Valid inequalities for the single-item capacitated lot sizing problem with step-wise costs. European Journal of Operational Research 198 (2), 412–434.
- [17] Akbalik, A., Rapine, C., 2012. Polynomial time algorithms for the constant capacitated singleitem lot sizing problem with stepwise production cost. Operations Research Letters 40 (5), 390–397.

- [18] Akbalik, A., Rapine, C., 2013. The single item uncapacitated lot-sizing problem with timedependent batch sizes: NP-hard and polynomial cases. European Journal of Operational Research 229 (2), 353–363.
- [19] Aksen, D., 2007. Loss of customer goodwill in the uncapacitated lot-sizing problem. Computers & operations research 34 (9), 2805–2823.
- [20] Aksen, D., Altinkemer, K., Chand, S., 2003. The single-item lot-sizing problem with immediate lost sales. European Journal of Operational Research 147 (3), 558–566.
- [21] Aloulou, M. A., Dolgui, A., Kovalyov, M. Y., 2014. A bibliography of non-deterministic lot-sizing models. International Journal of Production Research 52 (8), 2293–2310.
- [22] Alp, O., Erkip, N. K., Güllü, R., 2003. Optimal Lot-Sizing/Vehicle-Dispatching Policies Under Stochastic Lead Times and Stepwise Fixed Costs. Operations Research 51 (1), 160–166.
- [23] Anderson, E. J., Cheah, B. S., 1993. Capacitated lot-sizing with minimum batch sizes and setup times. International Journal of Production Economics 30-31 (0), 137 – 152, special Issue Proceeding of the Seventh International Working Seminar on Production Economics.
- [24] Anily, S., Tzur, M., 2005. Shipping multiple items by capacitated vehicles. Transportation Science 39 (2), 233–248.
- [25] Aouam, T., Brahimi, N., 2013. Integrated production planning and order acceptance under uncertainty: A robust optimization approach. European Journal of Operational Research 228 (3), 504–515.
- [26] Archetti, C., Bertazzi, L., Grazia Speranza, M., 2014. Polynomial cases of the economic lot sizing problem with cost discounts. European Journal of Operational Research 237 (2), 519–527.
- [27] Archetti, C., Bertazzi, L., Laporte, G., Speranza, M., 2007. A branch-and-cut algorithm for a vendor-managed inventory-routing problem. Transportation Science 41 (3), 382–391.
- [28] Archetti, C., Bertazzi, L., Paletta, G., Speranza, M. G., 2011. Analysis of the maximum level policy in a production-distribution system. Computers & Operations Research 38 (12), 1731– 1746.
- [29] Arkin, E., Joneja, D., Roundy, R., 1989. Computational complexity of uncapacitated multiechelon production planning problems. Operations Research Letters 8 (2), 61–66.
- [30] Armentano, V., Shiguemoto, A., Løkketangen, A., 2011. Tabu search with path relinking for an integrated production-distribution problem. Computers & Operations Research 38 (8), 1199– 1209.
- [31] Askin, R., 1983. The effect of lot-sizing on workload variability. Journal of Operations Management 4 (1), 53–71.
- [32] Atamtürk, A., Hochbaum, D. S., 2001. Capacity acquisition, subcontracting, and lot sizing. Management Science 47 (8), 1081.
- [33] Atamtürk, A., Küçükyavuz, S., 2005. Lot sizing with inventory bounds and fixed costs: Polyhedral study and computation. Operations Research 53 (4), 711–730.
- [34] Atamtürk, A., Küçükyavuz, S., 2008. An o (n2) algorithm for lot sizing with inventory bounds and fixed costs. Operations Research Letters 36 (3), 297–299.
- [35] Atamtürk, A., Muñoz, J. C., 2004. A study of the lot-sizing polytope. Mathematical Programming 99 (3), 443–465.

- [36] Baciarello, L., D'Avino, M., Onori, R., Schiraldi, M. M., 2013. Lot sizing heuristics performance. International Journal of Engineering Business Management 5 (6), 1–10.
- [37] Bahl, H. C., Neelam, B., 2009. An empirical comparison of lot-sizing methods available in an erp system. California Journal of Operations Management 7 (1), 77–83.
- [38] Bahl, H. C., Ritzman, L. P., Gupta, J. N. D., 1987. OR Practice-Determining lot sizes and resource requirements: A review. Operations Research 35 (3), 329–345.
- [39] Bahl, H. C., Taj, S., 1991. A data-dependent efficient implementation of the wagner-whitin algorithm for lot-sizing. Computers & Industrial Engineering 20 (2), 289 291.
- [40] Barany, I., Van Roy, T., Wolsey, L., 1984. Uncapacitated lot-sizing: The convex hull of solutions. In: Korte, B., Ritter, K. (Eds.), Mathematical Programming at Oberwolfach II. Vol. 22 of Mathematical Programming Studies. Springer Berlin Heidelberg, pp. 32–43.
- [41] Bard, J. F., Nananukul, N., 2010. A branch-and-price algorithm for an integrated production and inventory routing problem. Computers & Operations Research 37 (12), 2202–2217.
- [42] Ben-Tal, A., El Ghaoui, L., Nemirovski, A., 2009. Robust optimization. Princeton University Press.
- [43] Benders, J. F., 1962. Partitioning procedures for solving mixed-variables programming problems. Numerische mathematik 4 (1), 238–252.
- [44] Benli, O., Sabuncuoglu, I., Tufekci, S., 1988. A heuristic procedure for a single-item dynamic lot sizing problem. Computers & Industrial Engineering 14 (2), 181–192.
- [45] Beraldi, P., Ruszczyński, A., 2002. A branch and bound method for stochastic integer problems under probabilistic constraints. Optimization Methods and Software 17 (3), 359–382.
- [46] Berk, E., Toy, A. O., Hazir, O., 2008. Single item lot-sizing problem for a warm/cold process with immediate lost sales. European Journal of Operational Research 187 (3), 1251–1267.
- [47] Bertazzi, L., Paletta, G., Speranza, M., 2005. Minimizing the total cost in an integrated Vendor-Managed inventory system. Journal of Heuristics 11 (5-6), 393–419.
- [48] Bertsimas, D., Thiele, A., 2006. A robust optimization approach to inventory theory. Operations Research 54 (1), 150–168.
- [49] Bhattacharjee, S., Ramesh, R., 2000. A multi-period profit maximizing model for retail supply chain management: An integration of demand and supply-side mechanisms. European Journal of Operational Research 122 (3), 584–601.
- [50] Bitran, G. R., Matsuo, H., 1986. Approximation formulations for the single-product capacitated lot size problem. Operations Research 34 (1), 63–74.
- [51] Bitran, G. R., Yanasse, H. H., 1982. Computational complexity of the capacitated lot size problem. Management Science 28 (10), 1174.
- [52] Blackburn, J. D., Millen, R. A., 1980. Heuristic lot-sizing performance in a rolling-schedule environment. Decision Sciences 11 (4), 691–701.
- [53] Bollapragada, R., Rao, U. S., 2006. Replenishment planning in discrete-time, capacitated, nonstationary, stochastic inventory systems. IIE Transactions 38 (7), 605–617.
- [54] Bookbinder, J. H., Tan, J.-Y., 1988. Strategies for the probabilistic lot-sizing problem with service-level constraints. Management Science 34 (9), 1096–1108.

- [55] Boudia, M., Louly, M. A. O., Prins, C., 2007. A reactive GRASP and path relinking for a combined production-distribution problem. Computers & Operations Research 34 (11), 3402– 3419.
- [56] Brahimi, N., 2004. Production planning: New lot-sizing models and algorithms. Ph.D. thesis, PhD Thesis, Université de Nantes.
- [57] Brahimi, N., Absi, N., Dauzère-Pérès, S., Kedad-Sidhoum, S., 2015. Models and lagrangian heuristics for a two-level lot-sizing problem with bounded inventory. OR Spectrum 37 (4), 983– 1006.
- [58] Brahimi, N., Aouam, T., Aghezzaf, E.-H., 2015. Integrating order acceptance decisions with flexible due dates in a production planning model with load-dependent lead times. International Journal of Production Research 53 (12), 3810–3822.
- [59] Brahimi, N., Dauzère-Pérès, S., 2015. A lagrangian heuristic for capacitated single item lot sizing problems. 4OR 13 (2), 173–198.
- [60] Brahimi, N., Dauzère-Pérès, S., Najid, N. M., Nordli, A., 2006. Single item lot sizing problems. European Journal of Operational Research 168 (1), 1–16.
- [61] Brahimi, N., Khan, S. A., 2014. Warehouse location with production, inventory, and distribution decisions: a case study in the lube oil industry. 4OR 12 (2), 175–197.
- [62] Buschkühl, L., Sahling, F., Helber, S., Tempelmeier, H., 2010. Dynamic capacitated lot-sizing problems: a classification and review of solution approaches. Or Spectrum 32 (2), 231–261.
- [63] Chan, L. M. A., Muriel, A., Shen, Z.-J., Simchi-Levi, D., 2002. On the effectiveness of zeroinventory-ordering policies for the economic lot-sizing model with a class of piecewise linear cost structures. Operations Research 50 (6), 1058–1067.
- [64] Chan, L. M. A., Shen, Z. J. M., Simchi-Levi, D., Swann, J. L., 2004. Coordination of Pricing and Inventory Decisions: A Survey and Classification. In: Simchi-Levi, D., Wu, S. D., Shen, Z.-J. (Eds.), Handbook of Quantitative Supply Chain Analysis. No. 74 in International Series in Operations Research & Management Science. Springer US, pp. 335–392.
- [65] Chan, L. M. A., Simchi-Levi, D., Swann, J., 2006. Pricing, production, and inventory policies for manufacturing with stochastic demand and discretionary sales. Manufacturing & Service Operations Management 8 (2), 149–168.
- [66] Chen, J.-M., Chen, L.-T., 2005. Pricing and production lot-size/scheduling with finite capacity for a deteriorating item over a finite horizon. Computers & Operations Research 32 (11), 2801– 2819.
- [67] Chen, S., Feng, Y., Lin, B., 2008. An algorithm for single-item economic lot-sizing problem with general inventory cost, non-decreasing capacity, and non-increasing setup and production cost. Operations Research Letters 36 (3), 300–302.
- [68] Chen, X., Hu, P., 2012. Joint pricing and inventory management with deterministic demand and costly price adjustment. Operations Research Letters 40 (5), 385–389.
- [69] Chen, X., Simchi-Levi, D., 2004. Coordinating inventory control and pricing strategies with random demand and fixed ordering cost: The finite horizon case. Operations Research 52 (6), 887–896.
- [70] Chen, X., Simchi-Levi, D., 2012. Pricing and inventory management. In: Philips, R., Ozalp, O. (Eds.), The Oxford Handbook of Pricing Management. Oxford University Press, USA, pp. 784–824.

- [71] Chen, X., Zhou, S. X., Chen, Y. F., 2011. Integration of Inventory and Pricing Decisions with Costly Price Adjustments. Operations Research 59 (5), 1144–1158.
- [72] Chew, E. P., Lee, C., Liu, R., 2009. Joint inventory allocation and pricing decisions for perishable products. International Journal of Production Economics 120 (1), 139–150.
- [73] Choudhary, D., Shankar, R., 2011. Modeling and analysis of single item multi-period procurement lot-sizing problem considering rejections and late deliveries. Computers & Industrial Engineering 61 (4), 1318–1323.
- [74] Choudhary, D., Shankar, R., 2014. A goal programming model for joint decision making of inventory lot-size, supplier selection and carrier selection. Computers & Industrial Engineering 71, 1–9.
- [75] Chu, C., Chu, F., Zhong, J., Yang, S., 2013. A polynomial algorithm for a lot-sizing problem with backlogging, outsourcing and limited inventory. Computers & Industrial Engineering 64 (1), 200–210.
- [76] Chu, F., Chu, C., 2007. Polynomial algorithms for single-item lot-sizing models with bounded inventory and backlogging or outsourcing. Automation Science and Engineering, IEEE Transactions on 4 (2), 233–251.
- [77] Chubanov, S., Kovalyov, M. Y., Pesch, E., 2006. An FPTAS for a single-item capacitated economic lot-sizing problem with monotone cost structure. Mathematical Programming 106 (3), 453.
- [78] Chubanov, S., Kovalyov, M. Y., Pesch, E., 2008. A single-item economic lot-sizing problem with a non-uniform resource: Approximation. European Journal of Operational Research 189 (3), 877–889.
- [79] Chubanov, S., Pesch, E., 2012. An FPTAS for the single-item capacitated economic lot-sizing problem with supply and demand. Operations Research Letters 40 (6), 445–449.
- [80] Chung, C.-S., Flynn, J., Lin, C.-H. M., 1994. An effective algorithm for the capacitated single item lot size problem. European Journal of Operational Research 75 (2), 427–440.
- [81] Coleman, B., McKnew, M. A., 1995. A quick and effective method for capacitated lot sizing with startup and reservation costs. Computers & Operations Research 22 (6), 641–653.
- [82] Coleman, B. J., 1992. A further analysis of variable demand lot-sizing techniques. Production and Inventory Management Journal; Third Quarter 33 (3), 19–24.
- [83] Dauzère-Pérès, S., Brahimi, N., Najid, N., Nordli, A., April 2002. The single-item lot sizing problem with time windows. Tech. Rep. 02/4/AUTO, Ecole des Mines de Nantes, France.
- [84] de Toledo, F. M. B., Shiguemoto, A. L., 2005. Lot-sizing problem with several production centers. Pesquisa Operacional 25 (3), 479–492.
- [85] Dellaert, N., Melo, M., 1996. Production strategies for a stochastic lot-sizing problem with constant capacity. European Journal of Operational Research 92 (2), 281–301.
- [86] Dellaert, N. P., Melo, M. T., 1996. Stochastic lot-sizing: Solution and heuristic methods. International Journal of Production Economics 46, 261–276.
- [87] Dellaert, N. P., Melo, M. T., 2003. Approximate solutions for a stochastic lot-sizing problem with partial customer-order information. European Journal of Operational Research 150 (1), 163–180.

- [88] DeMatteis, J., 1968. An economic lot-sizing technique, i: The part-period algorithm. IBM Systems Journal 7 (1), 30–38.
- [89] Deng, S., Yano, C. A., 2006. Joint production and pricing decisions with setup costs and capacity constraints. Management Science 52 (5), 741–756.
- [90] Denizel, M., Solyalı, O., Süral, H., 2010. Tight formulations for the two and three level serial lot-sizing problems. In: International Workshop on Lot Sizing, IWLS'2010. Gardanne, France, pp. 40–42.
- [91] Di Summa, M., Wolsey, L., 2010. Lot-sizing with stock upper bounds and fixed charges. SIAM Journal on Discrete Mathematics 24 (3), 853–875.
- [92] Diaz-Madroñero, M., Mula, J., Peidro, D., 2014. A review of discrete-time optimization models for tactical production planning. International Journal of Production Research 52 (17), 5171– 205.
- [93] Egri, P., Kis, T., Kovács, A., Váncza, J., 2014. An inverse economic lot-sizing approach to eliciting supplier cost parameters. International Journal of Production Economics 149, 80–88.
- [94] Elmaghraby, W., Keskinocak, P., 2003. Dynamic Pricing in the Presence of Inventory Considerations: Research Overview, Current Practices, and Future Directions. Management Science 49 (10), 1287–1309.
- [95] Erenguc, S., Aksoy, Y., 1990. A branch and bound algorithm for a single item nonconvex dynamic lot sizing problem with capacity constraints. Computers & Operations Research 17 (2), 199–210.
- [96] Erenguc, S. S., Tufekci, S., 1987. A branch and bound algorithm for a single-item multi-source dynamic lot sizing problem with capacity constraints. IIE Transactions 19 (1), 73–80.
- [97] Escalante, M. S., Marenco, J. L., del Carmen Varaldo, M., 2011. A polyhedral study of the singleitem lot-sizing problem with continuous start-up costs. Electronic Notes in Discrete Mathematics 37, 261–266.
- [98] Fazle Baki, M., Chaouch, B. A., Abdul-Kader, W., 2014. A heuristic solution procedure for the dynamic lot sizing problem with remanufacturing and product recovery. Computers & Operations Research 43, 225–236.
- [99] Federgruen, A., Heching, A., 1999. Combined Pricing and Inventory Control Under Uncertainty. Operations Research 47 (3), 454–475.
- [100] Federgruen, A., Lee, C.-Y., 1990. The dynamic lot size model with quantity discount. Naval Research Logistics (NRL) 37 (5), 707–713.
- [101] Feng, K., Rao, U. S., Raturi, A., 2011. Setting planned orders in master production scheduling under demand uncertainty. International Journal of Production Research 49 (13), 4007–4025.
- [102] Feng, Y., Chen, S., Kumar, A., Lin, B., 2011. Solving single-product economic lot-sizing problem with non-increasing setup cost, constant capacity and convex inventory cost in o(n log n) time. Computers & Operations Research 38 (4), 717–722.
- [103] Fleischhacker, A. J., Zhao, Y., 2011. Planning for demand failure: A dynamic lot size model for clinical trial supply chains. European Journal of Operational Research 211 (3), 496–506.
- [104] Fleischmann, B., 1990. The discrete lot-sizing and scheduling problem. European Journal of Operational Research 44, 337–348.
- [105] Florian, M., Klein, M., 1971. Deterministic production planning with concave costs and capacity constraints. Management Science 18, 12–20.

- [106] Florian, M., Lenstra, J., Rinnooy Kan, H., 1980. Deterministic production planning: Algorithms and complexity. Management Science 26, 669–679.
- [107] Friedman, Y., Hoch, Y., 1978. A dynamic lot-size model with inventory deterioration. INFOR 16, 183–188.
- [108] Gade, D., Küçükyavuz, S., 2011. A note on "lot-sizing with fixed charges on stocks: the convex hull". Discrete Optimization 8 (2), 385–392.
- [109] Gahm, C., Denz, F., Dirr, M., Tuma, A., 2016. Energy-efficient scheduling in manufacturing companies: a review and research framework. European Journal of Operational Research 248 (3), 744–757.
- [110] Gallego, G., van Ryzin, G., 1994. Optimal Dynamic Pricing of Inventories with Stochastic Demand over Finite Horizons. Management Science 40 (8), 999–1020.
- [111] Ganas, I., Papachristos, S., 2005. The single-product lot-sizing problem with constant parameters and backlogging: Exact results, a new solution, and all parameter stability regions. Operations Research 53 (1), 170–176.
- [112] Gavish, B., Johnson, R. E., 1990. Fully polynomial approximation scheme for single-product scheduling in a finite capacity facility. Operations Research 38 (1), 70–83.
- [113] Gebennini, E., Gamberini, R., Manzini, R., 2009. An integrated production-distribution model for the dynamic location and allocation problem with safety stock optimization. International Journal of Production Economics 122 (1), 286–304.
- [114] Geunes, J., Romeijn, H. E., Taaffe, K., 2006. Requirements Planning with Pricing and Order Selection Flexibility. Operations Research 54 (2), 394–401.
- [115] Geunes, J., Taaffe, K., Romeijn, H., 2002. Models for integrated production planning and order selection. In: Proceedings of the 2002 Industrial Engineering Research Conference (IERC). Orlando. pp. 1–6.
- [116] Ghare, P., Shrader, G., 1963. A model for exponentially decaying inventories. Journal of Industrial Engineering 14, 238–243.
- [117] Ghodsypour, S., O'Brien, C., 1998. A decision support system for supplier selection using an integrated analytic hierarchy process and linear programming. International Journal of Production Economics 56-57 (0), 199 – 212.
- [118] Gilbert, S. M., 1999. Coordination of pricing and multi-period production for constant priced goods. European Journal of Operational Research 114 (2), 330–337.
- [119] Golany, B., Yang, J., Yu, G., 2001. Economic lot-sizing with remanufacturing options. IIE Transactions 33 (11), 995–1003.
- [120] Guan, Y., 2011. Stochastic lot-sizing with backlogging: computational complexity analysis. Journal of Global Optimization 49 (4), 651–678.
- [121] Guan, Y., Ahmed, S., Miller, A. J., Nemhauser, G. L., 2006. On formulations of the stochastic uncapacitated lot-sizing problem. Operations Research Letters 34 (3), 241–250.
- [122] Guan, Y., Ahmed, S., Nemhauser, G. L., Miller, A. J., 2006. A branch-and-cut algorithm for the stochastic uncapacitated lot-sizing problem. Mathematical Programming 105 (1), 55–84.
- [123] Guan, Y., Liu, T., 2010. Stochastic lot-sizing problem with inventory-bounds and constant order-capacities. European Journal of Operational Research 207 (3), 1398–1409.

- [124] Guan, Y., Miller, A. J., 2008. Polynomial-Time Algorithms for Stochastic Uncapacitated Lot-Sizing Problems. Operations Research 56 (5), 1172–1183.
- [125] Guide, J. V., R., D., Jayaraman, V., Srivastava, R., 1999. Production planning and control for remanufacturing: a state-of-the-art survey. Robotics and Computer-Integrated Manufacturing 15 (3), 221–230.
- [126] Guillaume, R., Kobylański, P., Zieliński, P., 2012. A robust lot sizing problem with ill-known demands. Fuzzy Sets and Systems 206, 39–57.
- [127] Gupta, A., Maranas, C. D., 2003. Managing demand uncertainty in supply chain planning. Computers & Chemical Engineering 27 (8-9), 1219–1227, 2nd Pan American Workshop in Process Systems Engineering.
- [128] Gutiérrez, J., Puerto, J., Sicilia, J., 2004. The multiscenario lot size problem with concave costs. European Journal of Operational Research 156 (1), 162–182.
- [129] Gutiérrez, J., Sedeno-Noda, A., Colebrook, M., Sicilia, J., 2001. A new characterization for the dynamic lot size problem with bounded inventory. Computers and Operations Research 30 (3), 383–395.
- [130] Guu, S.-M., Zhang, A. X., 2003. The finite multiple lot sizing problem with interrupted geometric yield and holding costs. European Journal of Operational Research 145 (3), 635–644.
- [131] Haase, K., 1994. Lotsizing and Scheduling for Production Planning. Springer-Verlag, New York.
- [132] Hardin, J. R., Nemhauser, G. L., Savelsbergh, M. W. P., 2007. Analysis of bounds for a capacitated single-item lot-sizing problem. Computers & Operations Research 34 (6), 1721–1743.
- [133] Haugen, K. K., Løkketangen, A., Woodruff, D. L., 2001. Progressive hedging as a meta-heuristic applied to stochastic lot-sizing. European Journal of Operational Research 132 (1), 116–122.
- [134] Hellion, B., Mangione, F., Penz, B., 2012. A polynomial time algorithm to solve the single-item capacitated lot sizing problem with minimum order quantities and concave costs. European Journal of Operational Research 222 (1), 10–16.
- [135] Hellion, B., Mangione, F., Penz, B., 2014. Stability contracts between supplier and retailer: a new lot sizing model. International Journal of Production Research 53 (1), 1–12.
- [136] Hindi, K., 1995. Efficient solution of the single-item, capacitated lot-sizing problem with start-up and reservation costs. Journal of the Operational Research Society 46 (10), 1223–1236.
- [137] Holmbom, M., Segerstedt, A., 2014. Economic order quantities in production: From harris to economic lot scheduling problems. International Journal of Production Economics 155, 82–90.
- [138] Hsu, V., 2000. Dynamic economic lot size model with perishable inventory. Management Science. 46(8), 1159–1169.
- [139] Hsu, V. N., Lowe, T. J., 2001. Dynamic economic lot size models with period-pair-dependent backorder and inventory costs. Operations Research 49 (2), 316–321.
- [140] Hu, J., Munson, C. L., Silver, E. A., 2004. A modified silver-meal heuristic for dynamic lot sizing under incremental quantity discounts. The Journal of the Operational Research Society 55 (6), 671–673.
- [141] Huang, K., Ahmed, S., 2010. A stochastic programming approach for planning horizons of infinite horizon capacity planning problems. European Journal of Operational Research 200 (1), 74–84.

- [142] Huang, K., Küçükyavuz, S., 2008. On stochastic lot-sizing problems with random lead times. Operations Research Letters 36 (3), 303–308.
- [143] Huang, L., Zhong, J.-H., Ni, Z.-W., Yang, S.-L., 2008. Polynomial algorithm for capacitated lot sizing models with shortage and outsourcing. Journal of System Simulation 20 (11), 2800–2805.
- [144] Hwang, H.-C., 2007. Dynamic lot-sizing model with production time windows. Naval Research Logistics (NRL) 54 (6), 692–701.
- [145] Hwang, H.-C., 2007. An efficient procedure for dynamic lot-sizing model with demand time windows. Journal of Global Optimization 37 (1), 11–26.
- [146] Hwang, H.-C., Jaruphongsa, W., 2008. Dynamic lot-sizing model for major and minor demands. European Journal of Operational Research 184 (2), 711–724.
- [147] Hwang, H.-C., van den Heuvel, W., 2012. Improved algorithms for a lot-sizing problem with inventory bounds and backlogging. Naval Research Logistics (NRL) 59 (3-4), 244–253.
- [148] Hwang, H.-C., van den Heuvel, W., Wagelmans, A. P., 2013. The economic lot-sizing problem with lost sales and bounded inventory. IIE Transactions 45 (8), 912–924.
- [149] Jans, R., Degraeve, Z., 2008. Modeling industrial lot sizing problems: a review. International Journal of Production Research 46 (6), 1619–1643.
- [150] Jaruphongsa, W., Çetinkaya, S., Lee, C.-Y., 2004. Warehouse space capacity and delivery time window considerations in dynamic lot-sizing for a simple supply chain. International Journal of Production Economics 92 (2), 169–180.
- [151] Jiang, R., Guan, Y., 2011. An -time algorithm for the stochastic uncapacitated lot-sizing problem with random lead times. Operations Research Letters 39 (1), 74–77.
- [152] Kaminsky, P., Simchi-Levi, D., 2003. Production and distribution lot sizing in a two stage supply chain. IIE Transactions 35 (11), 1065–1075.
- [153] Kaminsky, P., Swaminathan, J. M., 2004. Effective Heuristics for Capacitated Production Planning with Multiperiod Production and Demand with Forecast Band Refinement. Manufacturing & Service Operations Management 6 (2), 184–194.
- [154] Karimi, B., Fatemi Ghomi, S. M. T., Wilson, J. M., 2003. The capacitated lot sizing problem: a review of models and algorithms. Omega 31 (5), 365–378.
- [155] Karmarkar, U. S., Kekre, S., Kekre, S., 1987. The dynamic lot-sizing problem with startup and reservation costs. Operations Research 35 (3), 389.
- [156] Ketsarapong, S., Punyangarm, V., Phusavat, K., Lin, B., 2012. An experience-based system supporting inventory planning: A fuzzy approach. Expert Systems with Applications 39 (8), 6994–7003.
- [157] Khouja, M., Goyal, S., 2008. A review of the joint replenishment problem literature: 1989-2005.
 European Journal of Operational Research 186 (1), 1 16.
- [158] Kian, R., Ülkü Gürler, Berk, E., 2014. The dynamic lot-sizing problem with convex economic production costs and setups. International Journal of Production Economics 155 (0), 361 – 379, celebrating a century of the economic order quantity model.
- [159] Kimes, S. E., 1989. Yield management: A tool for capacity-considered service firms. Journal of Operations Management 8 (4), 348–363.

- [160] Koca, E., Yaman, H., Aktürk, M. S., 2014. Lot sizing with piecewise concave production costs. INFORMS Journal on Computing 26 (4), 767–779.
- [161] Koh, S. C. L., Saad, S. M., Jones, M. H., 2002. Uncertainty under mrp-planned manufacture: Review and categorization. International Journal of Production Research 40 (10), 2399–2421.
- [162] Kovalyov, M., Pesch, E., 2014. An o(n log n) algorithm for a single-item capacitated lot-sizing problem with linear costs and no backlogging. International Journal of Production Research 52 (13), 3758–3761.
- [163] Küçükyavuz, S., 2011. Mixed-integer optimization approaches to deterministic and stochastic inventory management. INFORMS TutORials in Operations Research (ed. J. Geunes) 8, 90–105.
- [164] Kuhn, H., 1997. A dynamic lot sizing model with exponential machine breakdowns. European Journal of Operational Research 100 (3), 514 – 536.
- [165] Kunreuther, H., Schrage, L., 1973. Joint Pricing and Inventory Decisions for Constant Priced Items. Management Science 19 (7), 732–738.
- [166] Küçükyavuz, S., Pochet, Y., 2009. Uncapacitated lot sizing with backlogging: the convex hull. Mathematical Programming 118 (1), 151–175.
- [167] Lasserre, J.-B., 1992. An integrated model for job-shop planning and scheduling. Management Science 38 (8), 1201–1211.
- [168] Lee, C., Çetinkaya, S., Wagelmans, A., 2001. A dynamic lot-sizing model with demand time windows. Management Science 47 (10), 1384–1395.
- [169] Lee, C.-Y., 2004. Inventory replenishment model: lot sizing versus just-in-time delivery. Operations Research Letters 32 (6), 581 – 590.
- [170] Lee, C. Y., Cetinkaya, S., Jaruphongsa, W., 2003. A dynamic model for inventory lot sizing and outbound shipment scheduling at a third-party warehouse. Operations Research 51 (5), 735–747.
- [171] Lei, L., Liu, S., Ruszczynski, A., Park, S., 2006. On the integrated production, inventory, and distribution routing problem. IIE Transactions 38 (11), 955–970.
- [172] Levi, R., Geunes, J., Romeijn, H. E., Shmoys, D. B., 2005. Inventory and facility location models with market selection. In: Integer Programming and Combinatorial Optimization. Springer, pp. 111–124.
- [173] Levi, R., Roundy, R. O., Shmoys, D. B., 2006. Primal-dual algorithms for deterministic inventory problems. Mathematics of Operations Research 31 (2), 267–284.
- [174] Levi, R., Shi, C., 2013. Approximation algorithms for the stochastic lot-sizing problem with order lead times. Operations Research 61 (3), 593–602.
- [175] Li, C., Liu, F., Cao, H., Wang, Q., 2009. A stochastic dynamic programming based model for uncertain production planning of re-manufacturing system. International Journal of Production Research 47 (13), 3657–3668.
- [176] Li, Y., Nemhauser, G., Savelsbergh, M., 2012. Pricing for production and delivery flexibility in single-item lot-sizing. Computers & Operations Research 39 (12), 3408–3419.
- [177] Liberopoulos, G., Tsikis, I., Delikouras, S., 2010. Backorder penalty cost coefficient "b": What could it be? International Journal of Production Economics 123 (1), 166 – 178.
- [178] Lippman, S. A., 1969. Optimal inventory policy with multiple set-up costs. Management Science 16 (1), 118–138.

- [179] Liu, T., 2008. Economic lot sizing problem with inventory bounds. European journal of operational research 185 (1), 204–215.
- [180] Liu, X., Chu, F., Chu, C., Wang, C., 2007. Lot sizing with bounded inventory and lost sales. International Journal of Production Research 45 (24), 5881–5894.
- [181] Liu, X., Tu, Y., 2008. Production planning with limited inventory capacity and allowed stockout. International Journal of Production Economics 111 (1), 180–191.
- [182] Loparic, M., Marchand, H., Wolsey, L. A., 2003. Dynamic knapsack sets and capacitated lotsizing. Mathematical Programming 95 (1), 53–69.
- [183] Loparic, M., Pochet, Y., Wolsey, L. A., 2001. The uncapacitated lot-sizing problem with sales and safety stocks. Mathematical Programming 89 (3), 487–504.
- [184] Lotfi, V., Yoon, Y.-S., 1994. Algorithm for the single-item capacitated lot-sizing problem with concave production and holding costs. Journal of the Operational Research Society 45 (8), 934– 941.
- [185] Love, S., 1973. Bounded production and inventory models with piecewise concave costs. Management Science 20(3), 313–318.
- [186] Melo, R. A., Wolsey, L. A., 2010. Uncapacitated two-level lot-sizing. Operations Research Letters 38 (4), 241 – 245.
- [187] Merzifonluoglu, Y., Geunes, J., Romeijn, H., 2007. Integrated capacity, demand, and production planning with subcontracting and overtime options. Naval Research Logistics (NRL) 54 (4), 433– 447.
- [188] Minner, S., 2009. A comparison of simple heuristics for multi-product dynamic demand lotsizing with limited warehouse capacity. International Journal of Production Economics 118 (1), 305–310.
- [189] Mirmohammadi, S. H., Shadrokh, S., Kianfar, F., 2009. An efficient optimal algorithm for the quantity discount problem in material requirement planning. Computers & Operations Research 36 (6), 1780–1788.
- [190] Moqri, M., Javadi, M., Yazdian, A., 2011. Supplier selection and order lot sizing using dynamic programming. International Journal of Industrial Engineering Computations 2 (2), 319–328.
- [191] Mula, J., Poler, R., García-Sabater, J., Lario, F., 2006. Models for production planning under uncertainty: A review. International Journal of Production Economics 103 (1), 271 – 285.
- [192] Nahmias, S., 1982. Perishable inventory theory: A review. Operations Research 30, 680–708.
- [193] Ng, C., Kovalyov, M. Y., Cheng, T., 2010. A simple FPTAS for a single-item capacitated economic lot-sizing problem with a monotone cost structure. European Journal of Operational Research 200 (2), 621–624.
- [194] Nydick, R. L., Weiss, H. J., 1989. An evaluation of variable-demand lot-sizing techniques. Production and Inventory Management Journal 30 (4), 41.
- [195] Okhrin, I., Richter, K., 2011. The linear dynamic lot size problem with minimum order quantity. International Journal of Production Economics 133 (2), 688–693.
- [196] Okhrin, I., Richter, K., 2011. An o(t3) algorithm for the capacitated lot sizing problem with minimum order quantities. European Journal of Operational Research 211 (3), 507–514.

- [197] Onal, M., 2016. The two-level economic lot sizing problem with perishable items. Operations Research Letters 44 (3), 403–408.
- [198] Onal, M., Romeijn, H. E., Sapra, A., van den Heuvel, W., 2015. The economic lot-sizing problem with perishable items and consumption order preference. European Journal of Operational Research 244 (3), 881–891.
- [199] Onal, M., van den Heuvel, W., Liu, T., 2012. A note on "the economic lot sizing problem with inventory bounds". European Journal of Operational Research 223 (1), 290–294.
- [200] Oral, M., Salvador, M. S., Reisman, A., Dean, B. V., 1972. On the evaluation of shortage costs for inventory control of finished goods. Management Science 18 (6), B-344-B-351.
- [201] Ou, J., 2012. Economic lot sizing with constant capacities and concave inventory costs. Naval Research Logistics (NRL) 59 (7), 497–501.
- [202] Pahl, J., Voß, S., 2014. Integrating deterioration and lifetime constraints in production and supply chain planning: A survey. European Journal of Operational Research 238 (3), 654 – 674.
- [203] Pai, P.-F., 2003. Capacitated Lot size problems with fuzzy capacity. Mathematical and Computer Modelling 38 (5–6), 661–669.
- [204] Pan, Z., Tang, J., Liu, O., 2009. Capacitated dynamic lot sizing problems in closed-loop supply chain. European Journal of Operational Research 198 (3), 810–821.
- [205] Park, Y. W., Klabjan, D., 2015. Lot sizing with minimum order quantity. Discrete Applied Mathematics 181 (0), 235 – 254.
- [206] Parsopoulos, K., Konstantaras, I., Skouri, K., 2015. Metaheuristic optimization for the singleitem dynamic lot sizing problem with returns and remanufacturing. Computers & Industrial Engineering 83 (0), 307 – 315.
- [207] Perakis, G., Zaretsky, M., 2008. Multiperiod Models with Capacities in Competitive Supply Chain. Production and Operations Management 17 (4), 439–454.
- [208] Pereira, O., Wolsey, L., 2001. On the wagner-whitin lot-sizing polyhedron. Mathematics of Operations Research 26 (3), 591–600.
- [209] Piperagkas, G. S., Konstantaras, I., Skouri, K., Parsopoulos, K. E., 2012. Solving the stochastic dynamic lot-sizing problem through nature-inspired heuristics. Computers & Operations Research 39 (7), 1555–1565.
- [210] Pochet, Y., Wolsey, L. A., 1993. Lot-sizing with constant batches: Formulation and valid inequalities. Mathematics of Operations Research 18 (4), 767–785.
- [211] Pochet, Y., Wolsey, L. A., 2006. Production planning by mixed integer programming. Springer, New York; Berlin.
- [212] Pujawan, I. N., Silver, E. A., 2008. Augmenting the lot sizing order quantity when demand is probabilistic. European Journal of Operational Research 188 (3), 705–722.
- [213] Pujawan, N., 2004. The effect of lot sizing rules on order variability. European Journal of Operational Research 159 (3), 617–635.
- [214] Rajagopalan, S., 1992. Deterministic capacity expansion under deterioration. Management Science 38, 525–539.

- [215] Retel Helmrich, M. J., Jans, R., van den Heuvel, W., Wagelmans, A. P., 2015. The economic lotsizing problem with an emission capacity constraint. European Journal of Operational Research 241 (1), 50 – 62.
- [216] Retel Helmrich, M. J., Jans, R., van den Heuvel, W., Wagelmans, A. P. M., 2014. Economic lot-sizing with remanufacturing: complexity and efficient formulations. IIE Transactions 46 (1), 67–86.
- [217] Richter, K., Sombrutzki, M., 2000. Remanufacturing planning for the reverse wagner/whitin models. European Journal of Operational Research 121 (2), 304–315.
- [218] Richter, K., Weber, J., 2001. The reverse wagner/whitin model with variable manufacturing and remanufacturing cost. International Journal of Production Economics 71 (1), 447–456.
- [219] Robinson, P., Narayanan, A., Sahin, F., 2009. Coordinated deterministic dynamic demand lotsizing problem: A review of models and algorithms. Omega 37 (1), 3 – 15.
- [220] Ruokokoski, M., Solyali, O., Cordeau, J.-F., Jans, R., Sural, H., 2010. Efficient formulations and a branch-and-cut algorithm for a production-routing problem. GERAD technical report g-2010-66, HEC Montréal, Canada.
- [221] Sana, S. S., Goyal, S. K., Chaudhuri, K., 2007. An imperfect production process in a volume flexible inventory model. International Journal of Production Economics 105 (2), 548–559.
- [222] Schulz, T., 2011. A new silver-meal based heuristic for the single-item dynamic lot sizing problem with returns and remanufacturing. International Journal of Production Research 49 (9), 2519– 2533.
- [223] Sedeño-Noda, A., Gutiérrez, J., Abdul-Jalbar, B., Sicilia, J., 2004. An o(t log t) algorithm for the dynamic lot size problem with limited storage and linear costs. Computational Optimization and Applications 28 (3), 311–323.
- [224] Senoussi, A., Mouss, N. K., Penz, B., Brahimi, N., Dauzère-Pérès, S., 2015. Modeling and solving a one-supplier multi-vehicle production-inventory-distribution problem with clustered retailers. The International Journal of Advanced Manufacturing Technology, 1–19.
- [225] Şenyiğit, E., Düğenci, M., Aydin, M. E., Zeydan, M., Mar. 2013. Heuristic-based neural networks for stochastic dynamic lot sizing problem. Applied Soft Computing 13 (3), 1332–1339.
- [226] Shiguemoto, A., Armentano, V., 2010. A tabu search procedure for coordinating production, inventory and distribution routing problems. International Transactions in Operational Research 17 (2), 179–195.
- [227] Shiue, Y.-C., 1990. Decision criteria on the branch-and-bound method for optimal single-level lot sizing with backlogging. Engineering Costs and Production Economics 20 (2), 139–150.
- [228] Silver, E., 1978. Inventory control under a probabilistic time-varying, demand pattern. A I I E Transactions 10 (4), 371–379.
- [229] Silver, E., Meal, H., 1973. A heuristic for selecting lot size quantities for the case of a deterministic time-varying demand rate and discrete opportunities for replenishment. Production and Inventory Management 14, 64–74.
- [230] Simchi-Levi, D., Chen, X., Bramel, J., 2014. Integration of Inventory and Pricing. In: The Logic of Logistics. Springer Series in Operations Research and Financial Engineering. Springer New York, pp. 177–209.

ACCEPTED MANUSCRIPT

- [231] Sodhi, M. S., 2005. Managing Demand Risk in Tactical Supply Chain Planning for a Global Consumer Electronics Company. Production and Operations Management 14 (1), 69–79.
- [232] Solyalı, O., Süral, H., 2012. The one-warehouse multi-retailer problem: reformulation, classification, and computational results. Annals of Operations Research 196 (1), 517–541.
- [233] Sox, C. R., 1997. Dynamic lot sizing with random demand and non-stationary costs. Operations Research Letters 20 (4), 155–164.
- [234] Stadtler, H., 2000. Improved rolling schedules for the dynamic single-level lot-sizing problem. Management Science 46 (2), 318–326.
- [235] Stevenson, W. J., 2014. Operations Management, 12th Edition. McGraw-Hill Education.
- [236] Swoveland, G., 1975. A deterministic multi-period production planning model with piecewise concave production and holding-backorder costs. Management Science 21 (9), 1007 1013.
- [237] Tan, C. H., Hartman, J. C., 2011. Sensitivity analysis in markov decision processes with uncertain parameters. Journal of Applied Probability 48 (4), 954–967.
- [238] Tang, O., Grubbström, R. W., 2002. Planning and replanning the master production schedule under demand uncertainty. International Journal of Production Economics 78 (3), 323–334.
- [239] Tarim, S., Kingsman, B. G., 2004. The stochastic dynamic production/inventory lot-sizing problem with service-level constraints. International Journal of Production Economics 88 (1), 105– 119.
- [240] Taskin, S., Lodree Jr., E. J., 2010. Inventory decisions for emergency supplies based on hurricane count predictions. International Journal of Production Economics 126 (1), 66–75.
- [241] Tempelmeier, H., 2007. On the stochastic uncapacitated dynamic single-item lotsizing problem with service level constraints. European Journal of Operational Research 181 (1), 184–194.
- [242] Tempelmeier, H., 2013. Stochastic Lot Sizing Problems. In: Smith, J. M., Tan, B. (Eds.), Handbook of Stochastic Models and Analysis of Manufacturing System Operations. No. 192 in International Series in Operations Research & Management Science. Springer New York, pp. 313–344.
- [243] Tempelmeier, H., Herpers, S., 2011. Dynamic uncapacitated lot sizing with random demand under a fillrate constraint. European Journal of Operational Research 212 (3), 497–507.
- [244] Teunter, R. H., Bayindir, Z. P., van den Heuvel, W., 2006. Dynamic lot sizing with product returns and remanufacturing. International Journal of Production Research 44 (20), 4377–4400.
- [245] Thomas, J., 1970. Price-production decisions with deterministic demand. Management Science 16 (11), 747–750.
- [246] Thomas, J., 1974. Price and production decisions with stochastic demand. Operations Research 22, 513–518.
- [247] Toczylowski, E., 1995. An o(t²) algorithm for the lot-sizing problem with limited inventory levels. In: Proceedings of the INRIA/IEEE Symposium on Emerging Technologies and Factory Automation (ETFA'95). Vol. 3. pp. 77–85.
- [248] Topaloglu, H., 2005. An approximate dynamic programming approach for a product distribution problem. IIE Transactions 37 (8), 697–710.
- [249] Toy, A. O., Berk, E., 2006. Dynamic lot sizing problem for a warm/cold process. IIE Transactions 38 (11), 1027–1044.

- [250] Tunc, H., Kilic, O. A., Tarim, S. A., Eksioglu, B., 2014. A reformulation for the stochastic lot sizing problem with service-level constraints. Operations Research Letters 42 (2), 161–165.
- [251] Vachani, R., 1992. Performance of heuristics for the uncapacitated lot-size problem. Naval Research Logistics (NRL) 39 (6), 801–813.
- [252] van den Heuvel, W., Kundakcioglu, O. E., Geunes, J., Romeijn, H. E., Sharkey, T. C., Wagelmans, A. P., M, 2012. Integrated market selection and production planning: complexity and solution approaches. Mathematical Programming 134 (2), 395–424.
- [253] van den Heuvel, W., Wagelmans, A. P., 2006. An efficient dynamic programming algorithm for a special case of the capacitated lot-sizing problem. Computers & Operations Research 33 (12), 3583 – 3599, part Special Issue: Recent Algorithmic Advances for Arc Routing Problems.
- [254] van den Heuvel, W., Wagelmans, A. P., 2008. Four equivalent lot-sizing models. Operations Research Letters 36 (4), 465–470.
- [255] van den Heuvel, W., Wagelmans, A. P. M., 2006. A polynomial time algorithm for a deterministic joint pricing and inventory model. European Journal of Operational Research 170 (2), 463–480.
- [256] van den Heuvel, W., Wagelmans, A. P. M., 2009. A holding cost bound for the economic lot-sizing problem with time-invariant cost parameters. Operations Research Letters 37 (2), 102–106.
- [257] van den Heuvel, W., Wagelmans, A. P. M., 2010. Worst-case analysis for a general class of online lot-sizing heuristics. Operations research 58 (1), 59–67.
- [258] van Eijl, C., van Hoesel, C., 1997. On the discrete lot-sizing and scheduling problem with wagner-whitin costs. Operations Research Letters 20 (1), 7–13.
- [259] van Hoesel, C. P. M., Wagelmans, A. P. M., Wolsey, L. A., 1994. Polyhedral characterization of the economic lot-sizing problem with start-up costs. SIAM Journal on Discrete Mathematics 7 (1), 11.
- [260] van Hoesel, S., Kuik, R., Salomon, M., van Wassenhove, L. N., 1994. The single item discrete lot-sizing and scheduling problem: Optimization by linear and dynamic programming. Discrete Applied Mathematics 48 (3), 289–303.
- [261] van Hoesel, S., Romeijn, H. E., Morales, D. R., Wagelmans, A. P. M., 2005. Integrated lot sizing in serial supply chains with production capacities. Management Science 51 (11), 1706–1719.
- [262] van Hoesel, S., Wagelmans, A., Kolen, A., 1991. A dual algorithm for the economic lot-sizing problem. European Journal of Operational Research 52 (3), 315–325.
- [263] van Vyve, M., 2006. Linear-programming extended formulations for the single-item lot-sizing problem with backlogging and constant capacity. Mathematical Programming 108 (1), 53–77.
- [264] van Vyve, M., 2007. Algorithms for single-item lot-sizing problems with constant batch size. Mathematics of Operations Research 32 (3), 594–613.
- [265] van Vyve, M., Ortega, F., 2004. Lot-sizing with fixed charges on stocks: the convex hull. Discrete Optimization 1 (2), 189–203.
- [266] Vargas, V., 2009. An optimal solution for the stochastic version of the Wagner–Whitin dynamic lot-size model. European Journal of Operational Research 198 (2), 447–451.
- [267] Vargas, V., Metters, R., 2011. A master production scheduling procedure for stochastic demand and rolling planning horizons. International Journal of Production Economics 132 (2), 296–302.

- [268] Wagner, H. M., 1960. A postscript to "dynamic problems in the theory of the firm". Naval Research Logistics Quarterly 7 (1), 7–12.
- [269] Wagner, H. M., Whitin, T. M., 1958. Dynamic version of the economic lot size model. Management Science 5 (1), 89.
- [270] Wagner, M. R., 2011. Online lot-sizing problems with ordering, holding and shortage costs. Operations Research Letters 39 (2), 144–149.
- [271] Wang, C.-H., Sheu, S.-H., 2000. Fast approach to the optimal production/PM policy. Computers & Mathematics with Applications 40 (10–11), 1297–1314.
- [272] Wang, J.-L., 2009. A supply chain application of fuzzy set theory to inventory control models DRP system analysis. Expert Systems with Applications 36 (5), 9229–9239.
- [273] Wang, N., He, Z., Sun, J., Xie, H., Shi, W., 2011. A single-item uncapacitated lot-sizing problem with remanufacturing and outsourcing. Proceedia Engineering 15, 5170–5178.
- [274] Wang, X., Tang, W., 2009. Optimal production run length in deteriorating production processes with fuzzy elapsed time. Computers & Industrial Engineering 56 (4), 1627–1632.
- [275] Wemmerlöv, U., 1989. The behavior of lot-sizing procedures in the presence of forecast errors. Journal of Operations Management 8 (1), 37–47.
- [276] Wemmerlöv, U., 1983. The part-period balancing algorithm and its look ahead-look back feature: A theoretical and experimental analysis of a single stage lot-sizing procedure. Journal of Operations Management 4 (1), 23 – 39.
- [277] Wolsey, L., 2006. Lot-sizing with production and delivery time windows. Mathematical Programming Ser. A 107, 471–489.
- [278] Wolsey, L. A., 1989. Uncapacitated lot-sizing problems with start-up costs. Operations Research 37 (5), 741.
- [279] Wolsey, L. A., 1995. Progress with single-item lot-sizing. European Journal of Operational Research 86 (3), 395–401.
- [280] Yan, H., Tang, S.-l., 2009. Pre-distribution and post-distribution cross-docking operations. Transportation Research Part E: Logistics and Transportation Review 45 (6), 843–859.
- [281] Yang, J., Golany, B., Yu, G., 2005. A concave-cost production planning problem with remanufacturing options. Naval Research Logistics (NRL) 52 (5), 443–458.
- [282] Yang, P., Wee, H., Chung, S., Ho, P., 2010. Sequential and global optimization for a closed-loop deteriorating inventory supply chain. Mathematical and Computer Modelling 52 (1-2), 161–176.
- [283] Yano, C. A., Lee, H. L., 1995. Lot sizing with random yields: A review. Operations Research 43 (2), 311.
- [284] Yilmaz, P., Çatay, B., 2006. Strategic level three-stage production distribution planning with capacity expansion. Computers & Industrial Engineering 51 (4), 609–620.
- [285] Zabel, E., 1964. Some generalizations of an inventory planning horizon theorem. Management Science 10, 465–471.
- [286] Zangwill, W. I., 1966. Production smoothing of economic lot sizes with non-decreasing requirements. Management Science 13 (3), 191.
- [287] Zangwill, W. I., 1967. The piecewise concave function. Management Science 13 (11), 900–912.

ACCEPTED MANUSCRIPT

- [288] Zangwill, W. I., 1969. A backlogging model and a multi-echelon model of a dynamic economic lot size production systema network approach. Management Science 15 (9), 506.
- [289] Zhang, M., 2011. Two-stage minimax regret robust uncapacitated lot-sizing problems with demand uncertainty. Operations Research Letters 39 (5), 342–345.
- [290] Zhang, M., Kucukyavuz, S., Yaman, H., 2012. A polyhedral study of multi-echelon lot sizing with intermediate demands. Operations Research 60(4), 918–935.
- [291] Zhang, Z.-H., Jiang, H., Pan, X., 2012. A lagrangian relaxation based approach for the capacitated lot sizing problem in closed-loop supply chain. International Journal of Production Economics 140 (1), 249–255.
- [292] Zhou, Z., Guan, Y., 2010. Stochastic lot-sizing problem with deterministic demands and Wagner–Whitin costs. Operations Research Letters 38 (5), 414–419.